Returns Volatility and Other Comprehensive Income Components

Dirk E. Black* Duke University dirk.black@duke.edu

January 2014

Abstract:

I examine whether other comprehensive income (OCI) component volatilities are associated with returns volatility for banks. I predict that OCI component volatilities have associations with returns volatility that vary in strength, and that inferences regarding the usefulness of OCI can be improved by analyzing associations between OCI component volatilities and returns volatility. I use returns volatility as the measure of total risk for a bank, consistent with the FASB's conceptual framework, and disaggregate OCI into its four primary components: available-forsale (AFS) securities adjustments; cash-flow hedge adjustments; pension-related adjustments; and foreign currency translation adjustments. Using hand-collected data, I further disaggregate AFS securities adjustments and cash-flow hedge adjustments into their unrealized and recycled subcomponents. I find that volatilities of unrealized (recycled) gains and losses on available-forsale securities and cash-flow hedges are negatively (positively) associated with returns volatility. I also find that associations between volatilities of these unrealized (recycled) gains and losses and returns volatility are more negative (stronger) when OCI is presented in a performance statement. The results indicate that volatilities of unrealized gains and losses, typically deemed beyond managers' control, are negatively associated with risk, while volatilities of recycled gains and losses, over which managers have relatively more control, are positively associated with risk.

*Correspondence information: 100 Fuqua Drive, Box 90120, Durham, NC 27708-0120, USA, tel: (919) 660-7957. I thank Ervin Black, Michael Bradbury, Shane Dikolli, Scott Dyreng, Peter Easton, Frank Ecker, Jennifer Francis, Amanda Gonzales, Ryan Kerr, Christian Lundblad, Per Olsson, Katherine Schipper, Thomas Steffen, Mohan Venkatachalam, Norman Wong, and workshop participants at Duke University, the AFAANZ Doctoral Symposium, and the Accounting PhD Rookie Recruiting and Research Camp for helpful conversations and comments. I thank the Deloitte Foundation for financial support and the Institute for Digital Research and Education (IDRE) at UCLA for providing helpful programming information. All errors are my own.

1. Introduction

In this study, I examine whether the volatilities of other comprehensive income (OCI) and its components are associated with information about the variability of investors' equity returns.¹ By examining this research question, I provide evidence on whether OCI and its components provide decision-useful information about the uncertainty, or variability, of investors' future cash flows.

Statement of Accounting Concepts No. 8 indicates that decision-useful information helps investors assess "the amount, timing, and uncertainty of (the prospects for) future net cash inflows" (FASB, 2010, p. 1-2). I follow the FASB conceptual framework and use time-series equity returns volatility as the benchmark for whether OCI component volatilities are associated with total risk for a firm (FASB, 2010; Ryan, 2012). I assume that investors efficiently impound risk-relevant information into equity share prices. Since OCI and its components are presented separately from net income under U.S. GAAP, I examine whether OCI volatility and OCI component volatilities are associated with returns volatility, controlling for net income volatility.

Prior work hypothesizes a positive association between returns volatility and financial statement line item volatility if the financial statement line item provides decision-useful information about risk. (Hodder et al., 2006; Khan and Bradbury, 2011; 2012). This "risk-relevance" hypothesis is supported for net income volatility and comprehensive income volatility (Hodder et al., 2006; Khan and Bradbury 2011; 2012). However, prior work has documented

¹ Prior work on comprehensive income and OCI focuses primarily on value relevance (Cheng, Cheung, and Gopalakrishnan, 1993; Dhaliwal, Subramanyam, and Trezevant, 1999; O'Hanlon and Pope, 1999; Cahan, Courtenay, Gronewoller, and Upton, 2000; Biddle and Choi, 2006; Chambers, Linsmeier, Shakespeare, and Sougiannis, 2007; Kanagaretnam, Mathieu, and Shehata, 2009; Barton, Hansen, and Pownall, 2010; Goncharov and Hodgson, 2011; Landsman, Miller, Peasnell, and Yeh, 2011, Dong, Ryan, and Zhang, 2013). However, the value relevance of financial statement line items is driven by persistence and risk (Lipe, 1986; Kormendi and Lipe, 1987; Easton and Zmijewski, 1989). Jones and Smith (2011) find that OCI is negatively persistent, while Hodder, Hopkins, and Wahlen (2006) and Khan and Bradbury (2011; 2012) provide mixed evidence on the relation between market-based risk proxies and the difference between comprehensive income volatility and net income volatility (incremental comprehensive income volatility).

statistically insignificant associations (p > 0.10) between returns volatility and the difference between comprehensive income volatility and net income volatility, hereafter referred to as incremental comprehensive income volatility (a proxy for OCI volatility) (Hodder et al., 2006; Khan and Bradbury 2011; 2012).² I predict that OCI component volatilities have associations with returns volatility that vary in strength, which may explain the lack of an association between OCI volatility and returns volatility. I test this prediction by disaggregating OCI in two ways.

First, I disaggregate OCI into its four primary or most frequently reported components: available-for-sale (AFS) securities adjustments; cash-flow hedge adjustments; pension-related adjustments; and foreign currency translation adjustments.³ I also include an "other" category for other OCI items reported by SNL Financial. AFS securities adjustments and cash-flow hedge adjustments derive primarily from changes in the fair values of AFS securities and cash-flow hedges. I classify these two components as "fair value" components. For the bank holding companies in my sample, these "fair value" components are the two largest and most reported OCI components, on average.

Pension-related adjustments arise primarily from differences between the projected benefit obligation and plan assets, differences between the expected and actual return on plan assets, prior service costs or credits, and transition assets or obligations. Foreign currency translation adjustments arise from the consolidation process, hedges of net investments in foreign companies, and gains and losses on long-term, within-firm foreign currency transactions. Since pension-related adjustments, foreign currency translation adjustments, and other OCI adjustments arise from a mixture of management estimates, actuarial assumptions, and the mechanical application of consolidation rules, I classify these components as "accounting

 $^{^{2}}$ I use the term "statistically significant" ("statistically insignificant") to indicate statistical significance (or lack thereof) of a result at the 0.10 level using a two-tailed test unless I indicate otherwise.

³ Appendix A lists OCI components (ASC 220-10-45-10A).

calculation" components. For the bank holding companies in my sample, these components are the three smallest and least reported OCI components, on average. I predict that associations between both "fair value" and "accounting calculation" component volatilities and returns volatility will be positive if the volatilities of OCI components generally reflect information also affecting investors' equity returns volatility. I also predict that OCI component volatilities' associations with returns volatility will vary by component.

I examine the relations between returns volatility and both "fair value" and "accounting calculation" OCI component volatilities using a sample of 2,264 annual bank holding company observations from 2002-2012 (the full sample). I find that the "fair value" component volatilities are not associated with returns volatility. For the "accounting calculation" components, I find that pension-related component volatility is negatively associated with returns volatility, though this negative association becomes statistically insignificant when I examine only observations with non-zero pension-related volatility. Foreign currency translation adjustment volatility is positively associated with returns volatility. I also find that OCI component volatilities have statistically distinct associations with returns volatility and are jointly significant in explaining variation in returns volatility.

The "fair value" OCI component volatilities may not appear to be associated with returns volatility because they are driven by the volatilities of both re-measurement (unrealized) gains and losses and realized (recycled) gains and losses that are transferred from accumulated other comprehensive income (AOCI) to net income.⁴ Volatilities of unrealized and recycled gains and losses on AFS securities and cash-flow hedges may reflect information also influencing investors' equity returns, in which case I would expect positive relations between unrealized

⁴ An example of a reclassification adjustment occurs when an unrealized gain on an AFS security from a prior period is reclassified to net income from AOCI upon sale of the AFS security.

subcomponent volatilities and returns volatility. Alternatively, volatilities of unrealized gains and losses may represent short-term fluctuations in the fair values of AFS securities and cash-flow hedges unrelated to risk, while volatilities of recycled gains and losses may represent sales of AFS securities or settlements of, cancellations of, or cessations of hedge accounting for cashflow hedge contracts unrelated to risk. I predict that associations between unrealized subcomponent volatilities and returns volatility are different from associations between recycled subcomponent volatilities and returns volatility, but make no directional prediction.

In my second approach, I hand-collect the unrealized and recycled OCI subcomponents related to AFS securities and cash-flow hedges from Forms 10-K and 10KSB (sec.gov) for a subsample of 898 observations and calculate their volatilities (the recycling sample).⁵ I find that the volatilities of unrealized gains and losses on AFS securities and cash-flow hedges are negatively associated with returns volatility, while the volatility of recycled gains and losses for AFS securities is positively associated with returns volatility.⁶

In sensitivity tests of approaches one and two, I rely on prior research that indicates financial statement users interpret OCI differently based on its presentation (Chambers et al., 2007; Hirst and Hopkins, 1998; Maines and McDaniel, 2000; Hirst, Hopkins and Wahlen, 2004; Hunton, Libby, and Mazza, 2006). I compare observations that use either of the performance statement presentation methods currently allowed under U.S. GAAP to observations that do not.⁷

⁵ I collect recycling data for only AFS securities and cash-flow hedge derivatives for three reasons. First, most banks provide a separation of the unrealized and realized portions of the AFS securities and cash-flow hedge derivative components of OCI. Second, many firms provide insufficient information to determine the following: (1) the amounts recycled to net income from the pension-related AOCI item; and (2) The amounts capitalized to an asset from the pension-related AOCI item. Third, there are relatively few instances of reclassification adjustments from foreign currency translation adjustments and the "other" category to net income.

⁶ The volatility of recycled gains and losses for cash-flow hedges is positively associated with returns volatility with p-values ranging from 0.101 to 0.121 depending on the estimation employed.

⁷ Following SFAS 130 (FASB, 1997), firms presented OCI in either a performance statement or in the statement of changes in equity. For fiscal years beginning after December 15, 2011, the option to present OCI in the statement of changes in equity was eliminated (FASB, 2011). Under current U.S. GAAP, a performance statement either begins

As prior work in this area provides mixed results regarding which presentation method investors understand better, I predict that associations between returns volatility and OCI component volatilities vary with presentation, but do not make a directional prediction.

I find no evidence that presentation affects associations between primary OCI component volatilities and returns volatility. However, when I disaggregate "fair value" component volatilities into their unrealized and recycled component volatilities, I find that OCI presentation affects the joint association between OCI component and subcomponent volatilities and returns volatility. I find that the positive (negative) associations between "fair value" recycled (unrealized) subcomponent volatilities and returns volatility are stronger (more negative) when OCI is presented in a performance statement.

Overall, I find evidence consistent with the prediction that associations between OCI component volatilities and returns volatility vary in strength. I find that these associations also vary in sign. The results indicate that the volatility of unrealized gains and losses, typically deemed beyond managers' control, is negatively associated with firm risk. On the other hand, the volatility of realized gains and losses, over which managers have relatively more control, is positively associated with firm risk.

The remainder of the paper is organized as follows: Section 2 motivates and presents the hypotheses. Section 3 presents sample information. Section 4 discusses research design and results. Section 5 discusses robustness checks. Section 6 concludes.

with revenue and ends with comprehensive income, or begins with net income and ends with comprehensive income. If a firm begins its performance statement with net income, the performance statement must immediately follow the income statement (ASC 220-10-45-1C).

2. Hypothesis Development

2.1 Returns Volatility, Comprehensive Income Volatility, and OCI Volatility

I follow the FASB conceptual framework and use investors' equity returns volatility as the benchmark for whether OCI component volatilities are associated with total risk for a firm. Returns volatility and OCI component volatilities may be directly associated with each other, or each may be associated with the information that drives the variability of equity returns. I assume that information about returns volatility is also information about the uncertainty of investors' future cash flows. Statement of Accounting Concepts No. 8 indicates that information that is decision-useful helps investors assess "the amount, timing, and uncertainty of (the prospects for) future net cash inflows" (FASB, 2010, p. 1-2). The conceptual framework also indicates that information about a firm's return to investors and "the variability and components of that return also is important, especially in assessing the uncertainty of future cash flows" (FASB, 2010, p. 4). Thus, if the volatility of an OCI component is positively associated with returns volatility, the OCI component reflects risk-relevant information.

Some opponents of the current labeling of OCI as part of an income or performance measure (comprehensive income) cite the volatility of OCI items as a reason that OCI items should not be referred to as "income," as this reference could be "confusing at best and potentially misleading" (Emerson, 2010, p. 2). One of the primary concerns about the exposure draft preceding SFAS 130 (Reporting Comprehensive Income) was that investors would draw incorrect inferences about firm risk based on the incremental volatility in OCI.⁸ In the Background Information and Basis for Conclusions section of SFAS 130, the FASB noted the concerns of several commenters on the standard:

⁸ Concerns about OCI volatility may arise due to incremental OCI volatility beyond that of net income, or because the current accounting standards for OCI allow firms to recognize fair value changes for "only selected economic assets, liabilities, and derivatives" (Hirst and Hopkins, 1998, p. 53).

"Some respondents indicated that comprehensive income would be volatile from period to period and that that volatility would be related to market forces beyond the control of management. In their view, therefore, it would be inappropriate to highlight that volatility in a statement of financial performance" (FASB, 1997, p. 21).

On the other hand, "Other respondents said that comprehensive income was more a measure of entity performance than it was of management performance and that it was therefore incorrect to argue that it should not be characterized as a performance measure because of management's inability to control the market forces that could result in that measure being volatile from period to period" (FASB, 1997, p. 21).

As noted in Graham, Harvey, and Rajgopal's (2005) report of survey results, "a few CFOs state that the market becomes more skeptical of underlying cash flows when earnings are volatile. Even if two firms have the same underlying cash flow volatility, executives believe that the firm with the more volatile earnings would be perceived as riskier" (Graham et al., 2005, p. 49). Managers also appear to believe that a firm with more volatile OCI could be assessed as "riskier" (FASB, 1997).

In prior work, the volatility of a financial statement line item has been measured as the firm-specific, time-series standard deviation of the item.⁹ For a sample of bank holding companies from 1996 to 2004, Hodder et al. (2006) provide evidence that comprehensive income (net income plus OCI) is more volatile than net income. Khan and Bradbury (2011) and Khan

⁹ For example, Barth, Landsman, and Wahlen (1995) estimate annual historical cost and fair value income per share volatilities by computing the standard deviations of these measures over the five-year period immediately preceding the observation of stock price at the end of the year for the years 1976-1990. Hodder et al. (2006, p. 352) estimate annual net income, comprehensive income, and full fair-value income volatilities by dividing each item by average total assets and calculating standard deviations of these measures "over five rolling five-year periods, each ending with years 2000-2004." Lee, Petroni, and Shen (2006, p. 681) estimate the relative volatility of comprehensive income as "the standard deviation of total comprehensive income divided by the standard deviation of net income using data from 1994-2001."

and Bradbury (2012) confirm this result for a sample of U.S. non-financial firms from 2005 to 2010 and a sample of New Zealand non-financial firms from 2003 to 2008. These authors hypothesize that if the volatilities of comprehensive income and OCI capture risk exposure, these volatilities should be positively associated with returns volatility. Hodder et al. (2006) and Khan and Bradbury (2011, 2012) each find a significant positive relation between comprehensive income volatility and returns volatility, suggesting that comprehensive income is risk relevant. As a proxy for OCI volatility, these authors deduct the volatility of net income from the volatility of comprehensive income and arrive at incremental comprehensive income volatility, a proxy for OCI volatility.¹⁰ Hodder et al. (2006) and Khan and Bradbury (2011; 2012) find that incremental comprehensive income volatility is not significantly related to returns volatility.

2.2 Returns Volatility and OCI Component Volatilities

I extend Hodder et al. (2006) and Khan and Bradbury (2011; 2012) in three ways. First, I consider both OCI volatility and OCI component volatilities. OCI component volatilities inform investors as to why summary performance measures, like comprehensive income, may be more or less volatile in a given fiscal period. The analyses in Hodder et al. (2006) and Khan and Bradbury (2011; 2012) constrain the associations between returns volatility and OCI component volatilities to be the same for all OCI components. OCI component volatilities could have correlations with returns volatility that vary in strength, which could confound inferences from analyzing only the association between OCI volatility and returns volatility.¹¹ Second, I consider

¹⁰ The variances, and thereby the standard deviations, of net income and OCI are not necessarily additive (Hodder et al., 2006, footnote 19, p. 360). Hodder et al. (2006, footnote 11, p. 351) indicate that the variance of comprehensive income is equal to: $\sigma_{CI}^2 = \sigma_{NI}^2 + \sigma_{OCI}^2 + 2Cov(NI, OCI)$. Hodder et al. (2006) calculate OCI volatility indirectly as $\sqrt{\sigma_{CI}^2} - \sqrt{\sigma_{NI}^2}$ and assume that Cov(NI, OCI) is equal to zero in their study. They provide some evidence consistent with this assumption for their sample of bank holding companies. For my sample, untabulated Pearson (-0.00, p = 0.83) and Spearman (-0.07, p < 0.05) correlation statistics between NI and OCI indicate that the covariance term in the expression above may not be equal to zero.

¹¹ Net income, a component of comprehensive income, may also have component volatilities with differing associations with risk. Lipe (1986) provides evidence that disaggregation of net income (a component of

additional factors that are associated with returns and may also be associated with returns volatility: size, growth opportunities, and firm performance. Third, I directly calculate OCI volatility (and OCI component volatilities), as opposed to using a calculated proxy for OCI volatility.

I examine bank holding companies because many, if not all, of the possible items that can be reported in OCI under U.S. GAAP are present for banks and because my empirical tests use the returns volatility model from Hodder et al. (2006). The primary components of OCI for most banks are AFS securities adjustments, cash-flow hedge adjustments, pension-related adjustments, and foreign currency translation adjustments. These components derive from a diverse set of economic events, management estimates, and mechanical accounting rule applications. I group these components into those that derive primarily from fair value changes ("fair value" OCI components) and those that derive primarily from a mixture of management estimates, actuarial assumptions, and mechanical applications of consolidation rules ("accounting calculation" OCI components). I predict that the distinct natures of these primary OCI components cause variation in the strength of their associations with returns volatility.

Associations between "fair value" component volatilities and returns volatility will be positive if the volatilities of the fair values of AFS securities and cash-flow hedge instruments generally reflect information also affecting investors' returns volatility. For the bank holding companies in my sample, these "fair value" components are the two largest and most reported OCI components, on average.

comprehensive income) provides incremental information to investors, and that this information varies with the persistence of net income components. Jones and Smith (2011) compare the value relevance, persistence, and predictive ability of special items (a net income component) and OCI. They find that special items and OCI are value relevant, that special items are not persistent and OCI is negatively persistent, and that special items have relatively better predictive ability for net income and cash flows than OCI. As my study is designed to inform the ongoing debate regarding the separation of OCI from net income, I provide evidence on whether OCI and its components are useful for risk assessment.

Pension-related adjustments arise from differences between the projected benefit obligation and plan assets, differences between the expected and actual return on plan assets, prior service costs or credits, transition assets or obligations, and amortization adjustments. Foreign currency translation adjustments arise from the consolidation process, hedges of net investments in foreign companies, and gains and losses on long-term, within-firm foreign currency transactions. I also include a category for "other" OCI components collected by SNL Financial. Pension-related adjustments, foreign currency translation adjustments, and other OCI adjustments arise from a heterogeneous mixture of management estimates, actuarial assumptions, and mechanical applications of consolidation rules. For the bank holding companies in my sample, these components are the three smallest and least reported OCI components, on average.

In Hypothesis 1, I follow prior research using the "risk-relevance" hypothesis and predict positive associations between returns volatility and OCI component volatilities. In Hypothesis 2, I compare OCI component volatilities' relative associations with returns volatility. <u>H1</u>: OCI component volatilities are positively associated with returns volatility. <u>H2</u>: OCI component volatilities have different associations with returns volatility.

2.3 Returns Volatility and OCI Unrealized and Recycled Subcomponent Volatilities

The volatilities of unrealized and recycled gains and losses on AFS securities and cashflow hedges may be associated with information simultaneously influencing investors' equity returns volatility, in which case I would expect positive relations between both unrealized and recycled subcomponent volatilities and returns volatility. Alternatively, associations between unrealized subcomponent volatilities and returns volatility could reflect "transitory" fluctuations in the fair values of AFS securities and cash-flow hedges that are not related to the risk of the firm (Dong et al., 2013), while associations between recycled subcomponent volatilities and

10

returns volatility could reflect fluctuations in sales of AFS securities or settlements of, cancellations of, or cessations of hedge accounting for cash-flow hedge contracts that are not related to the risk of the firm (Lee et al., 2006).

I predict that associations between unrealized subcomponent volatilities and returns volatility are different from associations between recycled subcomponent volatilities and returns volatility, but make no directional prediction.

<u>H3</u>: Unrealized OCI component volatilities differ from the volatilities of their recycled OCI counterpart volatilities in their associations with returns volatility.

3. Sample Selection

My initial sample consists of all U.S. bank holding company observations available on SNL Financial from 1998 to 2012. I begin the sample in 1998 since SFAS 130 is applicable for fiscal periods beginning after December 15, 1997 (FASB, 1997). Early research on OCI estimated OCI and its components using Compustat data (Dhaliwal et al., 1999). However, subsequent research suggests that using as-if estimates of OCI introduces measurement error (Chambers et al., 2007). For this reason, I use OCI data only from the post-SFAS-130 period and calculate rolling five-year standard deviations of OCI and OCI components beginning with 1998. Thus, the first estimation year included in the empirical tests is 2002.

Table 1 presents sample selection criteria. The sample from the post-SFAS-130 period (1998-2012) includes 10,034 bank-year observations from 757 banks. Eliminating observations prior to 2002, I obtain 7,828 observations from 754 banks. Requiring data for firm financial characteristics, comprehensive income components and their rolling five-year standard deviations, and primary regression control variables from SNL Financial per Hodder et al. (2006), I arrive at a sample of 6,962 bank-year observations from 728 banks. Including only

11

observations with 20 months of returns from CRSP to calculate rolling five-year returns volatility, I arrive at a sample of 2,264 bank-year observations from 298 banks (the full sample). Robustness tests that require data for "performance variables" – comprehensive income, lagged comprehensive income, annual stock returns, lagged annual stock returns, and lagged stock price – reduce the sample to 2,229 observations from 298 banks.

I also use the EDGAR database at sec.gov to access forms 10-K and 10KSB for bank holding companies to collect OCI recycling and presentation variables for a subsample of 898 observations from 121 banks (the recycling sample). Robustness tests that require data for comprehensive income, lagged comprehensive income, annual stock returns, lagged annual stock returns, and lagged stock price reduce the sample to 893 observations from 121 banks. I search for the word "comprehensive" to find the table presenting comprehensive income and OCI. I collect net-of-tax amounts recycled from AOCI to net income, along with the net-of-tax change in AOCI unrelated to recycling (i.e., the unrealized gain or loss) for AFS securities and cash-flow hedge derivatives. I also collect OCI presentation data and create an indicator variable equal to one when the bank presents OCI using one of the two currently allowed performance statement presentations (ASC 220-10-45-1C), zero otherwise.

Table 2 presents descriptive statistics for bank financial characteristics and regression variables used in the empirical tests.¹² All variables are described in Appendix B. Compared to descriptive statistics presented in Table 1 and Table 2 of Hodder et al. (2006), the banks in my full sample are larger based on average total assets, *AVGAT* (31,584 million vs. 15,746 million), book value of equity, *BVE* (2,794 million vs. 1,294 million), and market value of equity, *MVE*

¹² In Table 2, *AFS / OCI*, *DERIV / OCI*, *PEN / OCI*, *FC / OCI*, and *OTHER / OCI* are based on 2,246 observations because *OCI* is equal to zero for 18 observations from the full sample. *AFSUGL / OCI*, *AFSRECY / OCI*, *DERIVUGL / OCI*, and *DERIVRECY / OCI* are based on 897 observations because *OCI* is equal to zero for one observation from the recycling sample.

(3,439 million vs. 2,882 million), and similar based on deposits scaled by average total assets, DEP (79% vs. 79%), and OCI scaled by average total assets, OCI (0.01% vs. (1.149% - 1.133% = 0.02%)).¹³ Compared to the sample in Hodder et al. (2006), the banks in my sample are more volatile based on rolling five-year standard deviations of net income, σNI (0.55 vs. 0.26), comprehensive income, σCI (0.65 vs. 0.39), and OCI, σOCI (0.24 vs. (0.394 – 0.260 = 0.13)). Average comprehensive income volatility is greater than average net income volatility in my sample (0.65 vs. 0.55), confirming results from prior work (Hodder et al., 2006; Khan and Bradbury, 2011; 2012).¹⁴ In addition, the importance of directly computing OCI volatility is illustrated by the average difference between σCI and σNI , $\sigma CI - \sigma NI$ (0.10), versus average OCI volatility calculated directly, σOCI (0.24).¹⁵

Table 3 presents Pearson and Spearman correlation statistics for the primary regression variables. In this table only, I discuss statistical significance at the 0.05 level. I find significant positive Pearson and Spearman correlations between σCI and returns volatility (σRET), measured as the firm-specific rolling five-year standard deviation of monthly CRSP returns from year *t*-4 to year *t*, consistent with Hodder et al. (2006). I find a significant positive Pearson correlation between σOCI and σRET , but an insignificant negative Spearman correlation between σOCI and σRET . The Pearson and Spearman correlations between incremental comprehensive income volatility, $\sigma CI - \sigma NI$, and σRET are significantly negative, again illustrating the difference between incremental comprehensive income volatility and OCI

¹³ Firms appear to be increasing in size over my sample period based on average total assets and book value of equity. Since my sample period is 2002-2012, and that in Hodder et al. (2006) is 1996-2004, the increase in firm size over time likely explains the differences in these variables between the two samples.

¹⁴ An untabulated variance ratio test indicates that comprehensive income scaled by average total assets is significantly more volatile than net income scaled by average total assets.

¹⁵ Pearson (0.70, p < 0.05) and Spearman (0.60, p < 0.05) correlations between $\sigma CI - \sigma NI$ and σOCI suggest that these measures are not interchangeable. A t-test indicates that $\sigma CI - \sigma NI$ is significantly less than σOCI on average (t = -22.89, p < 0.01). Moreover, it is possible to obtain negative values of $\sigma CI - \sigma NI$, as indicated by the minimum value for $\sigma CI - \sigma NI$ in Table 2 (-0.65).

volatility. I find significant positive Pearson and Spearman correlations between three out of five OCI component volatilities ($\sigma DERIV$ – the cash-flow hedge component volatility, σFC – the foreign currency translation adjustment component volatility, and $\sigma OTHER$ – the "other" component volatility) and σRET , consistent with H1.

OCI volatility (σOCI) appears to be primarily driven by volatility in AFS securities adjustments (σAFS), as indicated by the large Pearson (0.97) and Spearman (0.91) correlations and the smaller correlations between σOCI and the other primary OCI component volatilities ($\sigma DERIV$, σPEN , σFC , and $\sigma OTHER$). σAFS is driven primarily by volatility in unrealized gains and losses on AFS securities ($\sigma AFSUGL$), as indicated by large Pearson (0.94) and Spearman (0.95) correlations.¹⁶ Thus, subsequent multivariate estimations of the relation between returns volatility and OCI volatility may be driven primary by the volatility of unrealized gains and losses on AFS securities. I also find that σRET is significantly positively correlated with the recycled subcomponent volatilities for AFS securities ($\sigma AFSRECY$) and cash-flow hedges ($\sigma DERICRECY$) (Pearson and Spearman). However, I find insignificant positive Pearson correlations between σRET and the volatilities of unrealized gains and losses on AFS securities ($\sigma AFSUGL$) and cash-flow hedges ($\sigma DERIVUGL$), a significant negative Spearman correlation between σRET and $\sigma AFSUGL$, and an insignificant positive correlation between σRET and $\sigma DERIVUGL$. These correlations indicate that associations between returns volatility and "fair value" OCI subcomponent volatilities vary based on whether gains and losses are unrealized or recycled, and provide support for H3.

¹⁶ I also find that $\sigma DERIV$ may be driven largely by volatility in unrealized gains and losses on cash-flow hedges ($\sigma DERIVUGL$), as indicated by large Pearson (0.97) and Spearman (0.96) correlations.

4. Research Design and Results

4.1 Returns Volatility, Comprehensive Income Volatility, and OCI Volatility

In this section, I compare results using my full sample to results from Hodder et al.

(2006). I investigate whether net income volatility (σNI), comprehensive income volatility (σCI), and incremental comprehensive income volatility ($\sigma CI - \sigma NI$) are associated with returns volatility. I then relax the assumption of variance additivity between net income and OCI and directly calculate OCI volatility (σOCI). Following Hodder et al. (2006), I estimate equation (1) using panel OLS regression with firm-clustered standard errors.

$$\sigma RET_{it} = \beta_0 + \beta_1 \sigma CI \ COMPONENT_{it} + \beta_2 EXP_{it} + \beta_3 GAP_{it} + \sum_{10} \gamma_{2002-2011} + \varepsilon_{it}$$
(1)

 σRET_{it} represents returns volatility for firm *i* in year *t*. $\sigma CI COMPONENT_{it}$ is the firmspecific rolling five-year standard deviation of either σNI_{it} or σCI_{it} calculated using fiscal years *t*-4 through *t*. I include two control variables, EXP_{it} and GAP_{it} , similar to Hodder et al. (2006). EXP_{it} represents firm *i*'s total exposure to derivatives in year *t*, proxied by total derivatives exposure from SNL Financial. GAP_{it} represents firm *i*'s exposure to interest rate changes in year *t*, proxied by the absolute value of the difference between interest-sensitive assets and interestsensitive liabilities scheduled to be re-priced within one year, per SNL Financial. I expect these control variables to be positively associated with returns volatility (Hodder et al., 2006). The model includes year fixed effects. All continuous independent variables are scaled by average total assets from SNL Financial.

Table 4, Columns (1) and (2) present the results of estimating equation (1). The signs of the coefficients on the two control variables, EXP_{it} and GAP_{it} , are consistent with Hodder et al. (2006) in each estimation, although the coefficients on GAP_{it} in the estimations in this paper are

significant, while those in Hodder et al. (2006) are not. Consistent with Hodder et al. (2006), I find significant positive associations between σRET_{it} and both σNI_{it} and σCI_{it} . Thus, both σNI_{it} and σCI_{it} appear to be risk-relevant based on their associations with returns volatility.¹⁷

Next, I estimate equation (2) to investigate the incremental risk relevance of OCI, controlling for net income volatility.

$$\sigma RET_{it} = \beta_0 + \beta_1 \sigma NI_{it} + \beta_2 \sigma OCIProxy_{it} + \beta_3 EXP_{it} + \beta_4 GAP_{it} + \sum_{10} \gamma_{2002-2011} + \varepsilon_{it} \quad (2)$$

 $\sigma OCIProxy_{it}$ equals incremental comprehensive income volatility, $\sigma CI_{it} - \sigma NI_{it}$, or OCI volatility, σOCI_{it} , calculated using rolling five-year standard deviations over fiscal years *t*-4 through *t*. Since my study concerns the risk-relevance of OCI components, rather than a comparison of the risk-relevance of net income, comprehensive income, and full fair value income as in Hodder et al. (2006), I do not collect data to estimate full fair value income volatility.¹⁸ Thus, equation (2) is not comparable with Hodder et al. (2006) because I do not include a measure of full fair value income volatility minus comprehensive income volatility (incremental full fair value income volatility).

Table 4, Columns (3) and (4) present the results of estimating equation (2). β_2 is interpreted as the incremental relevance of OCI volatility for returns volatility, controlling for net income volatility. I find that incremental comprehensive income volatility, $\sigma CI_{it} - \sigma NI_{it}$, is not significantly associated with returns volatility, consistent with Hodder et al.'s (2006) model

¹⁷ Estimating equation (1) for the recycling sample, I find results similar to those in Table 4, Columns (1) and (2), except that the coefficient on GAP_{it} is positive and insignificant. ¹⁸ I examine whether including the volatility of full fair value income minus comprehensive income in my empirical

¹⁸ I examine whether including the volatility of full fair value income minus comprehensive income in my empirical tests changes my primary inferences in Section 5.1.

which also includes incremental full fair value volatility.¹⁹ OCI volatility calculated directly (σOCI_{it}) is also not significantly associated with returns volatility.²⁰

4.2 Returns Volatility and OCI Component Volatilities

My first hypothesis predicts OCI component volatilities ($\sigma OCI COMPONENT_{it}$) are positively associated with returns volatility. I test H1 by calculating firm-specific rolling fiveyear OCI component volatilities over fiscal years *t*-4 to *t* for the "fair value" OCI components (σAFS_{it} – AFS securities component volatility and $\sigma DERIV_{it}$ – cash-flow hedge component volatility) and the "accounting calculation" OCI components (σPEN_{it} – pension-related component volatility, σFC_{it} – foreign currency translation adjustment volatility, and $\sigma OTHER_{it}$ – "other" component volatility) and estimate equation (3).

$$\sigma RET_{it} = \beta_0 + \beta_1 \sigma AFS_{it} + \beta_2 \sigma DERIV_{it} + \beta_3 \sigma PEN_{it} + \beta_4 \sigma FC_{it} + \beta_5 \sigma OTHER_{it} + \beta_6 \sigma NI_{it} + \sum_j \delta_j CONTROLS_{it} + \sum_{10} \gamma_{2002-2011} + \varepsilon_{it}$$
(3)

Table 5, Columns (1)-(3) present the results of estimating equation (3) using the full sample. Column (1) is the primary specification and uses the control variables from Hodder et al. (2006) (EXP_{it} , GAP_{it} , and year fixed effects). Column (2) adds controls for firm size and growth opportunities (MVE_{it} and BTM_{it}). Column (3) adds the performance controls (CI_{it} , CI_{it-1} , $ANNRET_{it}$, $ANNRET_{it-1}$, and $PRICE_{it-1}$). I find that the "fair value" component volatilities (σAFS_{it} and $\sigma DERIV_{it}$) are not significantly associated with returns volatility in Columns (1)-(3). Of the "accounting calculation" component volatilities, pension-related volatility (σPEN_{it}) is significantly associated with returns volatility, but its association is unexpectedly negative.

¹⁹ This finding is also consistent with Khan and Bradbury (2011; 2012) who examine samples of non-financial firms from the U.S. and New Zealand.

²⁰ Estimating equation (2) for the recycling sample (as in Table, 4, Columns (3) and (4)), I find significant negative coefficients on $\sigma CI_{it} - \sigma NI_{it}$ and σOCI_{it} and an insignificant positive coefficient on GAP_{it} .

Foreign currency translation adjustment volatility (σFC_{it}) is significantly positively associated with returns volatility.

A given firm may not have each component of OCI each year. Thus, I investigate whether the effects of OCI component volatilities in Table 5, Columns (1)-(3) are driven by observations that have non-zero values of OCI component volatilities. To do this, I estimate the specification in Table 5, Column (1) five additional times, each time restricting the sample to non-zero observations for a given OCI component volatility. I find that the significant negative association between pension-related volatility and returns volatility is no longer significant when I restrict the sample to observations with non-zero (positive) values of pension-related volatility. Thus, observations with zero values of pension-related volatility at least partially drive the initially-documented, significant negative relation between pension-related volatility and returns volatility. When I restrict the sample to observations with non-zero (positive) values of foreign currency translation adjustment volatility, I still find a positive significant association between foreign currency translation adjustment volatility and returns volatility. Inferences for the other primary OCI component volatilities are unaffected.

Table 5, Columns (4)-(6) presents the results of estimating equation (3) using the recycling sample. The results are similar to the full sample results in Columns (1)-(3) except that in Columns (4) and (5), I find negative and significant associations between the volatility of AFS securities adjustments (σAFS_{it}) and returns volatility. Thus, Table 5 provides no evidence in support of H1 for AFS securities adjustments, cash-flow hedge adjustments, pension adjustments, or "other" adjustments. Only the results for foreign currency translation adjustment volatility support H1.²¹

²¹ I examine whether OCI component volatilities reinforce or dampen each other's associations with returns volatility by examining each possible combination of two OCI component volatilities. For example, I estimate

My second hypothesis predicts OCI component volatilities have different associations with returns volatility. To test H2, I perform two joint hypothesis tests using the estimation results for equation (3) in each column. First, I test whether OCI component volatilities have different relative correlations with returns volatility by performing an F-test of coefficient equality ($\beta_1 = \beta_2 = \beta_3 = \beta_4 = \beta_5$) and find evidence in support of H2 for each column of Table 5. For example, in Table 5, Column (1), this F-test yields an F-statistic equal to 5.06, p < 0.01. Then, I test whether OCI component volatilities have different relative correlations with returns volatility and that these relative correlations are different from zero ($\beta_1 = \beta_2 = \beta_3 = \beta_4 = \beta_5 =$ 0). Again, I find evidence in support of H2 in each column of Table 5. For example, in Table 5, Column (1), this F-test yields an F-statistic equal to F = 4.28, p < 0.01. The results from Table 5 suggest that associations between returns volatility and OCI component volatilities vary not only in strength but also in sign.²²

4.3 Returns Volatility and OCI Unrealized and Recycled Subcomponent Volatilities

"Fair value" OCI components are composed of their unrealized and recycled subcomponents. The correlation matrix in Table 3 indicates σAFS_{it} and $\sigma DERIV_{it}$ are probably driven by volatility in their unrealized subcomponents, $\sigma AFSUGL_{it}$ and $\sigma DERIVUGL_{it}$. I estimate equation (4) including the "fair value" unrealized and recycled subcomponent volatilities and the "accounting calculation" component volatilities.

 $[\]sigma RET_{it} = \beta_0 + \beta_1 \sigma AFS_{it} + \beta_2 \sigma DERIV_{it} + \sigma NI_{it} + \beta_7 EXP_{it} + \beta_8 GAP_{it} + \sum_{10} \gamma_{2002-2011} + \varepsilon_{it}$ to see if $\sigma DERIV_{it}$ reinforces or dampens the association between σAFS_{it} and σRET_{it} . I find little evidence that OCI component volatilities significantly reinforce or dampen each other's associations with returns volatility using this modification of the "all OCI components" model presented in Table 5, Column (1).

²² To examine whether negative relations between returns volatility and OCI component volatilities occur only for bank holding companies, I examine relations between returns volatility and OCI component volatilities for a sample of non-financial firms from the Compustat/CRSP intersection from 2002 to 2012. I find significant negative associations between returns volatility and the following: $\sigma CI_{it} - \sigma NI_{it}$, σOCI_{it} , $\sigma DERIV_{it}$, σFC_{it} , and $\sigma OTHER_{it}$, indicating that negative associations between returns volatility and OCI component volatilities do not occur only for bank holding companies.

$$\sigma RET_{it} = \beta_0 + \beta_1 \sigma AFSUGL_{it} + \beta_2 \sigma AFSRECY_{it} + \beta_3 \sigma DERIVUGL_{it} + \beta_4 \sigma DERIVRECY_{it} + \beta_5 \sigma PEN_{it} + \beta_6 FC_{it} + \beta_7 \sigma THER_{it} + \beta_8 \sigma NI_{it} + \sum_j \delta_j CONTROLS_{it} + \sum_{10} \gamma_{2002-2011} + \varepsilon_{it}$$
(4)

Table 6, Columns (1)-(3) present the results of estimating equation (3). Column (1) is the primary specification and uses the control variables from Hodder et al. (2006) (EXP_{it}, GAP_{it}, and year fixed effects). Column (2) adds controls for firm size and growth opportunities (MVE_{it} and BTM_{it}). Column (3) adds the performance controls (CI_{it} , CI_{it-1} , $ANNRET_{it}$, $ANNRET_{it-1}$, and $PRICE_{it-1}$). I find that the volatilities of the unrealized gains and losses on AFS securities and cash-flow hedges ($\sigma AFSUGL_{it}$ and $\sigma DERIVUGL_{it}$) are significantly negatively associated with returns volatility, while the volatility of recycled gains and losses on AFS securities $(\sigma AFSRECY_{it})$ is significantly positively associated with returns volatility.²³ I also find that the volatility of recycled gains and losses on cash-flow hedges ($\sigma DERIVRECY_{it}$) is positively associated with returns volatility at the 0.115 level in Column (1), the 0.121 level in Column (2), and the 0.101 level in Column (3).²⁴ I test whether unrealized OCI subcomponent volatilities $(\sigma AFSUGL_{it} \text{ and } \sigma DERIVUGL_{it})$ differ from recycled OCI subcomponent volatilities $(\sigma AFSRECY_{it})$ and $\sigma DERIVRECY_{it}$ in their associations with returns volatility using F-tests for coefficient equality ($\beta_1 = \beta_2$ and $\beta_3 = \beta_4$) in each column. Both F-tests provide support for H3

²³ Net income is significantly positively associated with returns volatility in each column in Table 6. Since recycled gains and losses on AFS securities and cash-flow hedges are part of net income, I examine whether $\sigma AFSRECY_{it}$ and $\sigma DERIVRECY_{it}$ are positively associated with σNI_{it} by estimating $\sigma NI_{it} = \beta_0 + \beta_1 \sigma OCI \ COMPONENT_{it} + \beta_2 EXP_{it} + \beta_3 GAP_{it} + \sum_{10} \gamma_{2002-2011} + \varepsilon_{it}$. I find that $\sigma AFSRECY_{it}$ is significantly positively associated with σNI_{it} , while $\sigma DERIVRECY_{it}$ is insignificantly positively associated with σNI_{it} . The lack of association between $\sigma DERIVRECY_{it}$ and σNI_{it} may be driven by the relatively small number of non-zero values for $\sigma DERIVRECY_{it}$ (165) compared to $\sigma AFSRECY_{it}$ (799). ²⁴ If one-tailed t-tests were used, the p-values would be 0.0575, 0.0605, and 0.0505.

in each column. For example, in Column (1), the test of $\beta_1 = \beta_2$ yields an F-statistic equal to 5.67, p < 0.05; the test of $\beta_3 = \beta_4$ yields an F-statistic equal to 3.53, p < 0.10).²⁵

The evidence from Table 6 suggests that volatility in unrealized gains and losses of "fair value" OCI components is negatively associated with risk, while volatility in recycled gains and losses of "fair value" OCI components is positively associated with risk, similar to net income volatility.^{26, 27, 28}

4.4 Returns Volatility, OCI Component Volatilities, and Presentation

Prior to 2012, firms presented OCI in the statement of changes in equity, in a

performance statement beginning with net income and ending with comprehensive income, or in

a performance statement beginning with revenue and ending with comprehensive income. In

2012, the FASB eliminated the option to present OCI in the statement of changes in equity

²⁵ I also find evidence in support of H2 using the coefficients from the estimation of equation (4) in Table 6, Column (1)-(3) to perform F-tests of $\beta_1 = \beta_2 = \beta_3 = \beta_4 = \beta_5 = \beta_6 = \beta_7$ (Column (1): F = 3.33, p < 0.01; Column (2): F = 3.31, p < 0.01; Column (3): F = 3.40, p < 0.01) and $\beta_1 = \beta_2 = \beta_3 = \beta_4 = \beta_5 = \beta_6 = \beta_7 = 0$ (Column (1): F = 4.15, p < 0.01; Column (2): F = 4.18, p < 0.01; Column (3): F = 4.60, p < 0.01).

²⁶ I examine whether OCI component and subcomponent volatilities reinforce or dampen each other's associations with returns volatility by examining each possible combination of two OCI component volatilities. For example, I estimate $\sigma RET_{it} = \beta_0 + \beta_1 \sigma AFSUGL_{it} + \beta_2 \sigma AFSRECY_{it} + \sigma NI_{it} + \beta_7 EXP_{it} + \beta_8 GAP_{it} + \sum_{10} \gamma_{2002-2011} + \varepsilon_{it}$ to see if $\sigma AFSRECY_{it}$ reinforces or dampens the (insignificant) negative association between $\sigma AFSUGL_{it}$ and σRET_{it} . I find that $\sigma AFSRECY_{it}$ ($\sigma DERIVRECY_{it}$) reinforces the negative association between $\sigma AFSUGL_{it}$ ($\sigma DERIVUGL_{it}$) and σRET_{it} , while $\sigma AFSUGL_{it}$ ($\sigma DERIVUGL_{it}$) reinforces the positive association between $\sigma AFSRECY_{it}$ ($\sigma DERIVRECY_{it}$) and σRET_{it} , consistent with Table 6, using this modification of the "all OCI components" model presented in Table 6, Column (1). I also find that $\sigma AFSUGL_{it}$ reinforces the positive relation between σFC_{it} and σRET_{it} . However, the combined effects of (1) $\sigma AFSUGL_{it}$ and $\sigma AFSRECY_{it}$ and of (2) $\sigma DERIVUGL_{it}$ and $\sigma DERIVRECY_{it}$ each dampen the positive relation between σFC_{it} and σRET_{it} to the point of statistical insignificance, consistent with Table 6, Columns (1)-(3).

²⁷ To examine whether the associations between "fair value" OCI unrealized and recycled subcomponent volatilities and returns volatility arise in certain sample years, I re-estimate equation (4) from Table 6, Column (1) on an annual basis (excluding year fixed effects). The negative (positive) association between $\sigma AFSUGL_{it}$ ($\sigma AFSRECY_{it}$) and returns volatility derives primarily from the years 2008 to 2012 (2005, 2009 to 2012). The negative (positive) association between $\sigma DERIVUGL_{it}$ ($\sigma DERIVRECY_{it}$) and returns volatility derives primarily from the years 2002, 2003, 2005, and 2012.

²⁸ To examine whether the associations between "fair value" OCI subcomponent volatilities and returns volatility are sensitive to adjusting net income volatility to exclude recycling adjustments that transfer gains and losses from OCI to net income, I exclude recycling adjustments from net income and re-estimate equation (4) from Table 6, Column (1). My inferences are unchanged.

(FASB, 2011).²⁹ Prior research indicates financial statement users interpret OCI differently based on its presentation. Maines and McDaniel (2000) find that investors are better able to distinguish between high versus low volatility of unrealized gains and losses on AFS securities when OCI is presented in a performance statement than when it is presented in the statement of changes in equity.³⁰ Chambers et al. (2007) provide evidence, using realized returns as a dependent variable, that investors weight OCI most heavily when it is presented in a statement of changes in equity, the predominant presentation method in their sample of S&P 500 firms from 1998-2003, though for individual components, presentation method matters only for the pension component of OCI.

To examine whether the results in Tables 5 and 6 are sensitive to OCI presentation, I estimate equation (5) using the recycling sample for which I have also collected OCI presentation data.

$$\sigma RET_{it} = \beta_0 + \beta_1 \sigma AFS_{it} + \beta_2 \sigma DERIV_{it} + \beta_3 \sigma PEN_{it} + \beta_4 \sigma FC_{it} + \beta_5 \sigma OTHER_{it} + \beta_6 PRES_{it} + \beta_7 \sigma AFS_{it} * PRES_{it} + \beta_8 \sigma DERIV_{it} * PRES_{it} + \beta_9 \sigma PEN_{it} * PRES_{it} + \beta_{10} \sigma FC_{it} * PRES_{it} + \beta_{11} \sigma OTHER_{it} * PRES_{it} + \beta_{12} \sigma NI_{it} + \sum_j \delta_j CONTROLS_{it} + \sum_{10} \gamma_{2002-2011} + \varepsilon_{it}$$
(5)

²⁹ For fiscal years beginning after December 15, 2011 (FASB, 2011), comprehensive income and OCI may be presented in a performance statement in one of two forms, per ASC 220-10-45-1C: "A single continuous statement of comprehensive income or in a statement of net income and statement of other comprehensive income." If option two is elected, ASC 220-10-45-1B requires that the statement of other comprehensive income "be presented immediately after the statement of net income." In the empirical tests in this study, the presentation indicator variable equals one only when the firm uses either of the two currently allowed presentation methods. Thus, if the firm presents other comprehensive income in a performance statement that does not immediately follow the income statement, the indicator variable is set equal to zero.

³⁰ Bloomfield, Nelson, and Smith (2006) consider whether feedback loops between unrealized gains and losses on AFS securities and returns can cause volatility in equity prices in an experimental markets setting. Using MBA students as experimental participants, the authors find that price volatility is highest when firm investment in perfectly-correlated securities is high and when unrealized gains and losses are reported in a statement of comprehensive income. Koonce (2006) suggests that the subjects in Bloomfield et al. (2006) may not have had the ability to adjust their valuation decisions based on the correlation structure of investments. In addition, investors may be unable to observe the correlation structure of a firm's returns with its investment returns. Further, Koonce (2006) questions the frequency with which feedback loops would occur in real-world settings, citing insufficient investment in correlated securities, low correlations between a firm's returns and the returns on its investments, and immaterial amounts of unrealized gains and losses as potential threats to the external validity of the study.

PRES_{it} is an indicator variable equal to one if the bank reports OCI in either a single statement of comprehensive income or in a separate statement of other comprehensive income immediately following the income statement. This coding approach aligns with current FASB guidance for OCI presentation (ASC 220-10-45-1C). If the interactions between OCI component volatilities and presentation in equation (5) are significantly different from zero, presentation affects associations between OCI component volatilities and returns volatility. I also test whether presentation affects the joint association between OCI component volatilities and returns volatility using an F-test for the joint significance of the interactions terms in equation (5) $(\beta_7 = \beta_8 = \beta_9 = \beta_{10} = \beta_{11} = 0)$. Table 7, Columns (1)-(3) present the results of estimating equation (5). Column (1) is the primary specification and uses the control variables from Hodder et al. (2006) (EXP_{it}, GAP_{it}, and year fixed effects). Column (2) adds controls for firm size and growth opportunities (MVE_{it} and BTM_{it}). Column (3) adds the performance controls (CI_{it} , CI_{it-1} , $ANNRET_{it}$, $ANNRET_{it-1}$, and $PRICE_{it-1}$). I find no evidence that presentation affects individual or joint (i.e., Column (1): F = 0.61, p > 0.10) associations between OCI component volatilities and returns volatility for any estimation in Table 7.³¹

Next, I estimate equation (6) for the "fair value" OCI subcomponent volatilities $(\sigma AFSUGL_{it}, \sigma AFSRECY_{it}, \sigma DERIVUGL_{it}, \text{ and } \sigma DERIVRECY_{it})$ and expand the estimation of equation (5) to include these subcomponent volatilities in equation (6).

$$\sigma RET_{it} = \beta_0 + \beta_1 \sigma AFSUGL_{it} + \beta_2 \sigma AFSRECY_{it} + \beta_3 \sigma DERIVUGL_{it} + \beta_4 \sigma DERIVRECY_{it} + \beta_5 \sigma PEN_{it} + \beta_6 \sigma FC_{it} + \beta_7 \sigma OTHER_{it} + \beta_8 PRES_{it} + \beta_9 \sigma AFSUGL_{it} * PRES_{it} + \beta_{10} \sigma AFSRECY_{it} * PRES_{it} + \beta_{11} \sigma DERIVUGL_{it} * PRES_{it} + \beta_{12} \sigma DERIVRECY_{it} * PRES_{it} + \beta_{13} \sigma PEN_{it} * PRES_{it} + \beta_{14} \sigma FC_{it} * PRES_{it} + \beta_{15} \sigma OTHER_{it} * PRES_{it} + \beta_{16} \sigma NI_{it} + \sum_{j} \delta_j CONTROLS_{it} + \sum_{10} \gamma_{2002-2011} + \varepsilon_{it}$$
(6)

³¹ The coefficient estimates for σAFS_{it} , $\sigma DERIV_{it}$, σPEN_{it} , σFC_{it} , and $\sigma OTHER_{it}$ in Table 7, Columns (1)-(3) are consistent with Table 5, Columns (4)-(6).

Table 8, Columns (1)-(3) present the results of estimating equation (6). Column (1) is the primary specification and uses the control variables from Hodder et al. (2006) (EXP_{it}, GAP_{it}, and year fixed effects). Column (2) adds controls for firm size and growth opportunities (MVE_{it} and BTM_{it}). Column (3) adds the performance controls (CI_{it} , CI_{it-1} , $ANNRET_{it}$, $ANNRET_{it-1}$, and $PRICE_{it-1}$). I find that associations between the volatilities of unrealized gains and losses $(\sigma AFSUGL_{it} \text{ and } \sigma DERIVUGL_{it})$ and returns volatility are significantly more negative when OCI is presented in a performance statement in Columns (1) and (2).³² I also find that associations between the volatilities of recycled gains and losses ($\sigma AFSRECY_{it}$ and $\sigma DERIVRECY_{it}$) and returns volatility are significantly stronger when OCI is presented in a performance statement in Columns (1)-(3).³³ The coefficient on $\sigma DERIVRECY_{it} * PRES_{it}$ is large relative to the other coefficient estimates (i.e., Column (1): $\beta_{12} = 75.11$). Since some banks may not report cash-flow hedge adjustments in OCI in a given year, or in several consecutive years, it is possible that the large coefficient on $\sigma DERIVRECY_{it} * PRES_{it}$ is driven by a relatively small group of observations with non-zero values of $\sigma DERIVRECY_{it}$ that report OCI in a performance statement. Of the 272 observations (out of 898) with $PRES_{it} = 1$, 56 have nonzero values of $\sigma DERIVRECY_{it}$, potentially explaining why β_{12} is relatively large.³⁴ I also find consistent evidence in all three columns in Table 8 that presentation affects the joint association

 $^{33} \sigma AFSUGL_{it}$ and $\sigma DERIVUGL_{it}$ in Table 8, Column (1)-(3) are significantly negatively associated with returns volatility, similar to Table 6, Column (1)-(3). $\sigma AFSRECY_{it}$ in Table 8, Columns (1)-(3) is positively associated with returns volatility at the 0.125, 0.125, and 0.174 levels using two-tailed tests. These p-values would be 0.0625, 0.0625, and 0.087 using one-tailed tests. $\sigma DERIVRECY_{it}$ in Table 8, Columns (1)-(3) is positively associated with returns volatility at the 0.150, 0.163, and 0.141 levels using two-tailed tests. These p-values would be 0.075, 0.0815, and 0.0705 using one-tailed tests. In addition, F-tests presented at the bottom of Table 8, Columns (1)-(3) consistently indicate that both $\sigma AFSRECY_{it} + \sigma AFSRECY_{it} * PRES_{it}$ (p < 0.01) and $\sigma DERIVRECY_{it} + \sigma DERIVRECY_{it} * PRES_{it}$ (p < 0.05) are significantly associated with returns volatility.

³² In Column (6), the coefficients on $\sigma AFSUGL_{it} * PRES_{it}$ and $\sigma DERIVUGL_{it} * PRES_{it}$ are also negative, but are statistically insignificant.

³⁴ Of the 272 observations (out of 898) with $PRES_{it} = 1$, 99 have non-zero values of $\sigma DERIVUGL_{it}$, 263 have non-zero values of $\sigma AFSRECY_{it}$, and 272 have non-zero values of $\sigma AFSUGL_{it}$, mitigating concerns about the stability of the coefficients on $\sigma DERIVUGL_{it} * PRES_{it}$, $\sigma AFSREC_{it} * PRES_{it}$, and $\sigma AFSUGL_{it} * PRES_{it}$.

between OCI component volatilities and returns volatility from F-tests of $\beta_9 = \beta_{10} = \beta_{11} = \beta_{12} = \beta_{13} = \beta_{14} = \beta_{15} = 0$ (i.e., Column (1), F = 2.28, p < 0.05).

The results indicate that the volatility of unrealized gains and losses, typically deemed beyond managers' control, is negatively associated with risk, while the volatility of realized gains and losses, over which managers have relatively more control, is positively associated with risk.

5. Robustness Checks

In Sections 5.1-5.3, I explore the sensitivity of my inferences using alternative estimations of equation (6) from Table 8, Column (1) since this model links most closely with prior work per its control variables (Hodder et al., 2006), includes both the "fair value" subcomponent volatilities and "accounting calculation" component volatilities, and allows associations with returns volatility to vary with presentation. In Section 5.4, I examine the sensitivity of my inferences using alternative measures of σAFS_{it} and $\sigma DERIV_{it}$ to estimate equation (5) from Table 7, Column (1) since this model links most closely with prior work per its control variables (Hodder et al., 2006), includes all primary OCI component volatilities (σAFS_{it} , $\sigma DERIV_{it}$, σFEN_{it} , σFC_{it} , and $\sigma OTHER_{it}$), and allows associations with returns volatility to vary with presentation. In Section 5.5, I examine whether negative associations between volatilities of unrealized gains and losses ($\sigma AFSUGL_{it}$ and $\sigma DERIVUGL_{it}$) and σRET_{it} are driven by negative correlations between debt and equity market returns volatilities. In Section 5.6, I examine whether positive associations between volatilities of recycled gains and losses ($\sigma AFSRECY_{it}$ and $\sigma DERIVRECY_{it}$) and σRET_{it} are likely to reflect earnings smoothing.³⁵

³⁵ I do not further investigate the negative association between pension-related volatility and returns volatility because I previously demonstrated that the significance of this relation is statistically insignificant when I consider only observations with non-zero pension-related volatility ($\sigma PEN_{it} > 0$ for 502 out of 2,264 observations).

5.1 Additional Control Variables

Firm-specific characteristics other than derivatives exposure (EXP_{it}) and interest-rate gap (GAP_{it}) could be driving associations between OCI component volatilities and returns volatility. I address this concern in five ways. First, I include firm fixed effects in equation (6) from Table 8, Column (1). I use caution when interpreting these results because the large number of estimated parameters reduces the degrees of freedom in the model significantly and increases the probability of a multicollinearity problem. The coefficients on $\sigma AFSUGL_{it}$ and $\sigma DERIVUGL_{it}$ are negative and insignificant. However, F-tests indicate that both $\sigma AFSUGL_{it} + \sigma AFSUGL_{it} * PRES_{it}$ and $\sigma DERIVUGL_{it} + \sigma DERIVUGL_{it} * PRES_{it}$ are significantly associated with returns volatility. Since the coefficients on $\sigma AFSUGL_{it} * PRES_{it}$ and $\sigma DERIVUGL_{it}$ are negative and significant, I still conclude that the volatilities of unrealized gains and losses on available-for-sale securities and cash-flow hedges are negatively associated with returns volatility, and the associations between the volatilities of these unrealized gains and losses are more negative when OCI is presented in a performance statement.

Second, I control for returns volatility over the years *t*-9 to *t*-5 to control for autocorrelation in returns volatility.³⁶ Requiring an additional five years of returns (at least 20 additional monthly returns) reduces my recycling sample from 898 to 529 observations. I find a positive significant coefficient on $\sigma DERIVRECY_{it}$, a negative insignificant coefficient on σPEN_{it} , and do not find that associations between volatilities of unrealized and recycled gains and losses on cash-flow hedges are affected by OCI presentation, though the coefficient on $\sigma DERIVRECY_{it} * PRES_{it}$ is positive and significant at the 0.108 level. All other inferences are unchanged.

³⁶ I do not control for returns volatility over the years *t*-5 to *t*-1 because of the significant overlap such a control variable would have with the dependent variable, σRET_{it} , which is calculated over the years *t*-4 to *t*.

Third, I control for incremental full-fair-value income volatility ($\sigma(FFVI_{it} - CI_{it})$), using a full-fair-value income measure based on full-fair-value balance sheet changes (Blankespoor, Linsmeier, Petroni, and Shakespeare, 2013). My inferences are unchanged.

Fourth, I include the covariance of net income with each OCI component and subcomponent (e.g., COV(NI, AFS)) to control for the possibility that covariances of net income with OCI components and subcomponents may be causing negative associations between returns volatility and unrealized gains and losses on AFS securities and cash-flow hedges. I find a positive significant coefficient on $\sigma DERIVRECY_{it}$, a negative significant coefficient on $\sigma OTHER_{it}$, a negative insignificant coefficient on $\sigma AFSUGL_{it} * PRES_{it}$ (p = 0.128), and an insignificant F-statistic testing $\sigma AFSRECY_{it} + \sigma AFSRECY_{it} * PRES_{it} = 0$ (F = 2.71). All other inferences are unchanged.

Fifth, I control for liquidity using the rolling five-year average of the daily price impact of a trade (Amihud, 2002; Lang and Maffett, 2011) and the rolling five-year average of the bidask spread (Gow, Taylor, and Verrecchia, 2013). These liquidity controls are included because the negative relations between returns volatility and the volatilities of unrealized gains and losses on AFS securities and cash-flow hedges may reflect a relation between returns volatility and investors' trading liquidity. In the same model, I control for bank leverage using the rolling fiveyear average of total liabilities to average total assets (x 100) because investors in more levered banks may be more sensitive to OCI volatility than investors in less levered banks given the potential positive relation between risk and leverage (Khan and Bradbury, 2012). I find a positive significant coefficient on $\sigma AFSRECY_{it}$, a negative insignificant coefficient on $\sigma DERIVUGL_{it} * PRES_{it}$, a positive insignificant coefficient on $\sigma DERIVRECY_{it} * PRES_{it}$ (p = 0.133), and a positive and significant F-statistic testing $\sigma OTHER_{it} + \sigma OTHER_{it} * PRES_{it} = 0$ (F = 5.70). All other inferences are unchanged.

5.2 "Over Controlling"

Next, I examine whether my inferences are altered by removing control variables proposed by Hodder et al. (2006) because including total derivatives exposure (EXP_{it}) , interest rate gap (GAP_{it}), and year fixed effects may "over control" for information that drives associations between OCI component volatilities and returns volatility. Removing EXP_{it} and GAP_{it} from, and keeping year fixed effects in, equation (6), I find a negative insignificant coefficient on $\sigma DERIVUGL_{it}$ and a positive significant coefficient on σFC_{it} . All other inferences are unchanged. Removing year fixed effects from, and keeping EXP_{it} and GAP_{it} in, equation (6), I find a negative insignificant coefficient on $\sigma DERIVUGL_{it}$ and a negative insignificant coefficient on $\sigma DERIVUGL_{it} * PRES_{it}$. I also find a positive significant coefficient on $PRES_{it}$, likely due to time-clustering in the presentation variable that is controlled when year fixed effects are included in the model. All other inferences are unchanged. Removing EXP_{it}, GAP_{it}, and year fixed effects from equation (6), I find negative insignificant coefficients on $\sigma DERIVUGL_{it}$ and $\sigma DERIVUGL_{it} * PRES_{it}$ (p = 0.119). I find that $\sigma DERIVUGL_{it} + \sigma DERIVUGL_{it} * PRES_{it}$ is insignificantly different from zero (F = 2.54). I again find a positive significant coefficient on *PRES*_{*it*}. All other inferences are unchanged.

5.3 Alternative Variable Construction

Finally, I explore whether alternative constructions of returns volatility and OCI component volatilities affect my inferences. Instead of using the volatility of total returns (σRET_{it}) , I use the volatility of excess returns, calculated as the rolling five-year standard deviation of the bank's monthly return from CRSP minus the monthly return on five-year U.S.

Treasury bonds from the CRSP U.S. Treasury and Inflation Indexes dataset. I find that $\sigma OTHER_{it} + \sigma OTHER_{it} * PRES_{it}$ is significantly different from zero (F = 2.79). All other inferences are unchanged. Since I use rolling five-year standard deviations of OCI components as proxies for OCI component volatilities, it is possible that abnormal OCI amounts in a given year could affect OCI component volatilities for five sample years. To address this problem, I calculate OCI component volatilities using all post-SFAS 130 observations from 1998-2012. I then estimate equation (4) from Table 6, Column (1) without year fixed effects instead of equation (6) from Table 8, Column (1) because firms were only permitted to report OCI in a performance statement for fiscal years beginning after December 15, 2011 (FASB, 2011). I find a positive significant coefficient on $\sigma DERIVRECY_{it}$ and a negative insignificant coefficient on σPEN_{it} . All other inferences remain unchanged.

5.4 Alternative Measures of σAFS_{it} and $\sigma DERIV_{it}$

When hand-collecting the data for unrealized and recycled gains and losses on AFS securities and cash-flow hedges, I occasionally encounter observations where *AFSUGL* + *AFSRECY* or *DERIVUGL* + *DERIVRECY* are not equal to *AFS* and *DERIV* reported by SNL Financial. I re-estimate equation (5) from Table 7, Column (1) with volatilities calculated using my hand-collected data, where AFS = AFSUGL + AFSRECY and DERIV = DERIVUGL + DERIVRECY instead of *AFS* and *DERIV* from SNL Financial. I find a negative coefficient on σAFS_{it} significant at the 0.15 level. All other inferences remain unchanged.

5.5 Debt and Equity Market Returns Volatilities

Negative associations between debt and equity market returns volatilities could explain negative associations between σRET_{it} and both $\sigma AFSUGL_{it}$ and $\sigma DERIVUGL_{it}$ because both $\sigma AFSUGL_{it}$ and $\sigma DERIVUGL_{it}$ are tied to the debt market. Most banks hold large amounts of AFS debt securities as a proportion of their total AFS portfolios.³⁷ In addition, cash-flow hedges are often used to protect against interest-rate risk associated with forecasted transactions. I perform two tests to examine whether debt and equity market returns volatilities are negatively associated. First, I examine Pearson and Spearman correlations between σRET_{it} for my full sample and rolling five-year standard deviations of monthly returns on five-year U.S. Treasury bonds from the CRSP U.S. Treasury and Inflation Indexes dataset. I find positive significant Pearson and Spearman correlations equal to 0.09. Second, I examine the Pearson and Spearman correlations between rolling five-year standard deviations of monthly value-weighted S&P 500 returns from the CRSP Index File on the S&P 500 from WRDS between 2002-2012 and rolling five-year standard deviations of five-year U.S. Treasury bond returns over the same time period. I find positive insignificant Pearson (0.28) and Spearman correlations (0.11). Thus, it appears that negative associations between debt and equity market returns volatilities are unlikely to explain negative associations between volatilities of unrealized gains and losses and returns volatility.

5.6 Earnings Smoothing

Recycled gains (losses) simultaneously decrease (increase) OCI and increase (decrease) net income. Net income volatility is positively associated with returns volatility. Correlations in Table 3 and results mentioned in footnote 23 indicate that volatilities of recycled gains and losses are positively associated with net income volatility, although the association is stronger between $\sigma AFSRECY_{it}$ and σNI_{it} than between $\sigma DERIVRECY_{it}$ and σNI_{it} . Thus, positive associations between σRET_{it} and both $\sigma AFSRECY_{it}$ and $\sigma DERIVRECY_{it}$ could be the result of earnings

 $^{^{37}}$ I confirm this statement by obtaining AFS securities data for 1,799 of my 2,264 observations from the Bank Regulatory – Bank Holding Companies dataset on WRDS and calculating *AFSMIX* = (*AFSDEBT / AFSTOTAL*) * 100. I find that the mean (median) of the rolling five-year average of *AFSMIX* is 97.38% and 99.17% for these 1,799 observations.

smoothing, though managers have relatively less control over cash-flow hedge recycling because cash-flow hedge contracts typically hedge cash flows several years in the advance, while managers can sell AFS securities at any time. Using my recycling sample, I test for earnings smoothing in three ways.

First, prior work documents that negative associations between net income before recycled gains and losses and recycled gains and losses provide evidence of earnings smoothing (Lee et al., 2006). Since OCI recycled gains and loss are the opposite sign of recycled gains and losses recognized in net income, I test for the presence of earnings smoothing by examining whether *positive* associations exist between NI + AFSRECY (NI + DERIVRECY) and AFSRECY (DERIVRECY). I find an insignificant positive Pearson correlation between NI + AFSRECY and AFSRECY (0.02), a significant positive Spearman correlation between NI + AFSRECY and AFSRECY (0.09), and significant positive Pearson and Spearman correlations between NI + DERIVRECY and DERIVRECY (Pearson = 0.08, Spearman = 0.08). Second, if recycled gains and losses are used to smooth earnings, the volatility of net income should be less than the volatility of net income before recycled gains and losses. I compare the standard deviation of NI to the standard deviations of both NI + AFSRECY and NI + DERIVRECY and find no evidence that NI has a smaller standard deviation than either NI + AFSRECY (F = 1.02) or NI + DERIVRECY (F = 1.00). Third, I use two-sample t-tests and find that σNI is not less than either $\sigma(NI + AFSRECY)$ (t = 0.76) or $\sigma(NI + DERIVRECY)$ (t = 0.18). Overall, the evidence for earnings smoothing is either weak or non-existent for my sample.

6. Conclusion

I examine whether OCI components are associated with returns volatility for banks. I predict that OCI component volatilities have associations with returns volatility that vary in strength, and that inferences regarding the usefulness of OCI can be improved by analyzing associations between OCI component volatilities and returns volatility. I use returns volatility as the measure of total risk for a bank, consistent with the FASB's conceptual framework, and disaggregate OCI into its four primary components: AFS securities adjustments, cash-flow hedge adjustments; pension-related adjustments; and foreign currency translation adjustments. I find some evidence that pension-related OCI volatility is negatively associated with returns volatility, but this evidence weakens considerably when examining only observations with non-zero pension-related volatility. I also find some evidence of a positive association between foreign currency translation adjustment volatility and returns volatility, but the significance of this evidence is significantly dampened when "fair value" subcomponent volatilities are included in the same model as foreign currency translation adjustment volatility. I find no evidence of significant associations between the other primary OCI component volatilities and returns volatility.

Using hand-collected data, I further disaggregate AFS securities adjustments and cashflow hedge adjustments into their unrealized and recycled subcomponents. I find evidence that the volatilities of unrealized (recycled) gains and losses on AFS securities and cash-flow hedges are negatively (positively) associated with returns volatility. I also find that the associations between the volatilities of these unrealized (recycled) gains and losses and returns volatility are more negative (stronger) when OCI is presented in a performance statement. The results indicate that the volatility of unrealized gains and losses, typically deemed beyond managers' control, is

32

negatively associated with risk, while the volatility of realized gains and losses, over which managers have relatively more control, is positively associated with risk.

The results are robust to a variety of alternative empirical approaches. The negative associations between volatilities of unrealized gains and losses and returns volatility do not appear to be driven by negative associations between debt and equity market returns volatilities. The positive associations between volatilities of recycled gains and losses and returns volatility do not appear to be driven by earnings smoothing. The results are relevant for accounting standard setters seeking to understand potential characteristics that distinguish OCI components from net income and the relation between OCI volatility and the variability of investors' returns.

References

Amihud, Y., 2002, "Illiquidity and Stock Returns: Cross-section and Time-series Effects," *Journal of Financial Markets* 5, 31-56.

Barth, M., W. Landsman, and J. Wahlen, 1995, "Fair Value Accounting: Effects on Banks' Earnings Volatility, Regulatory Capital, and Value of Contractual Cash Flows," *Journal of Banking & Finance* 19, 577-605.

Barton, J., T. Hansen, and G. Pownall, 2010, "Which Performance Measures Do Investors Around the World Value the Most—and Why?" *The Accounting Review* 85, 753-789.

Biddle G. and J. Choi, 2006, "Is Comprehensive Income Useful?" *Journal of Contemporary Accounting & Economics 2*, 1-32.

Blankespoor, E., T. Linsmeier, K. Petroni, and C. Shakespeare, 2013, "Fair Value Accounting for Financial Instruments: Does It Improve the Association between Bank Leverage and Credit Risk?" *The Accounting Review* 88, 1143-1177.

Bloomfield, R., M. Nelson, and S. Smith, 2006, "Feedback Loops, Fair Value Accounting and Correlated Investments," *Review of Accounting Studies* 11, 377-416.

Cahan, S., S. Courtenay, P. Gronewoller, and D. Upton, 2000, "Value Relevance of Mandated Comprehensive Income Disclosures," *Journal of Business Finance & Accounting* 27, 1273-1301.

Chambers, D., T. Linsmeier, C. Shakespeare, and T. Sougiannis, 2007, "An Evaluation of SFAS No. 130 Comprehensive Income Disclosures," *Review of Accounting Studies* 12, 557-593.

Cheng, A. J. Cheung, and V. Gopalakrishnan, 1993, "On the Usefulness of Operating Income, Net Income, and Comprehensive Income in Explaining Equity Returns," *Accounting and Business Research* 23, 195-203.

Dhaliwal, D., K.R. Subramanyam, and R. Trezevant, 1999, "Is Comprehensive Income Superior to Net Income as a Measure of Firm Performance," *Journal of Accounting and Economics* 26, 43-67.

Dong, M., S. Ryan, and X. Zhang, 2013, "Preserving Amortized Costs within a Fair-Value-Accounting Framework: Reclassification of Gains and Losses on Available-for-Sale Securities upon Realization," *Review of Accounting Studies*, Forthcoming, Published online 28 August 2013.

Easton, P. and M. Zmijewski, 1989, "Cross-Sectional Variation in the Stock Market Response to Accounting Earnings Announcements," *Journal of Accounting and Economics* 11, 117-141.

Emerson, 2010, "Re: File Reference No. 1790-100, Comprehensive Income," Comment Letter No. 70.

Financial Accounting Standards Board (FASB), 1997, "Statement of Financial Accounting Standards No. 130: Reporting Comprehensive Income," Financial Accounting Foundation, Norwalk, CT.

Financial Accounting Standards Board (FASB), 2010, "Statement of Financing Accounting Concepts No. 8: Conceptual Framework for Financial Reporting," Financial Accounting Foundation, Norwalk, CT.

Financial Accounting Standards Board (FASB), 2011, "Accounting Standards Update No. 2011-05: Comprehensive Income (Topic 220)," Financial Accounting Foundation, Norwalk, CT.

Goncharov, I. and A. Hodgson, 2011, "Measuring and Reporting Income in Europe," *Journal of International Accounting Research* 10, 27-59.

Gow, I., D. Taylor, and R. Verrecchia, 2013, "On the Complementary Relation between Earnings and Private Information," Working Paper.

Graham, J., C. Harvey, and S. Rajgopal, 2005, "The Economic Implications of Corporate Financial Reporting," *Journal of Accounting Economics* 40, 3-73.

Hirst, D. and P. Hopkins, 1998, "Comprehensive Income Reporting and Analysts' Valuation Judgments," *Journal of Accounting Research* 36, 47-75.

Hirst, D.E., P. Hopkins, and J. Wahlen, 2004, "Fair Values, Income Measurement, and Bank Analysts' Risk and Valuation Judgments," *The Accounting Review* 79, 453-472.

Hodder, L., P. Hopkins, and J. Wahlen, 2006, "Risk relevance of Fair-Value Income Measures for Commercial Banks," *The Accounting Review* 81, 337-375.

Hunton, J., R. Libby, and C. Mazza, 2006, "Financial Reporting Transparency and Earnings Management," *The Accounting Review* 81, 135-157.

Jones, D. and K. Smith, 2011, "Comparing the Value Relevance, Predictive Value, and Persistence of Other Comprehensive Income and Special Items," *The Accounting Review* 86, 2047-2073.

Kanagaretnam, K., R. Mathieu, and M. Shehata, 2009, "Usefulness of Comprehensive Income Reporting in Canada," *Journal of Accounting and Public Policy* 28, 349-365.

Khan, S. and M. Bradbury, 2011, "Volatility and Risk relevance of Comprehensive Income," Working Paper.

Khan, S. and M. Bradbury, 2012, "Risk Relevance of Comprehensive Income," Working Paper.

Koonce, L., 2006, "Discussion of 'Feedback Loops, Fair Value Accounting and Correlated Investments," *Review of Accounting Studies* 11, 417-427.

Kormendi, R. and R. Lipe, 1987, "Earnings Innovations, Earnings Persistence, and Stock Returns," *The Journal of Business* 60, 323-345.

Landsman, W., B. Miller, K. Peasnell, and S. Yeh, 2011, "Do Investors Understand Really Dirty Surplus?" *The Accounting Review* 86, 237-258.

Lang, M. and M. Maffett, 2011, "Transparency and Liquidity Uncertainty in Crisis Periods," *Journal of Accounting and Economics* 52, 101-125.

Lee, Y., K. Petroni, and M. Shen, 2006, "Cherry Picking, Disclosure Quality, and Comprehensive Income Reporting Choices: The Case of Property-Liability Insurers," *Contemporary Accounting Research* 23, 655-692.

Lipe, R., 1986, "The Information Contained in the Components of Earnings," *Journal of Accounting Research* 24, 37-64.

Maines, L. and L. McDaniel, 2000, "Effects of Comprehensive-Income Characteristics on Nonprofessional Investors' Judgments: The Role of Financial-Statement Presentation Format," *The Accounting Review* 75, 179-207.

O'Hanlon, J. and P. Pope, 1999, "The Value-Relevance of UK Dirty Surplus Accounting Flows," *British Accounting Review* 31, 459-482.

Ryan, S., 2012, "Risk Reporting Quality: Implications of Academic Research for Financial Reporting Policy," *Accounting and Business Research* 42, 295-324.

Appendix A – Other Comprehensive Income (OCI) Components

Other Comprehensive Income Component (ASC 220-10-45-10A)	Reference Va				
Unrealized holdings gains and losses on available-for-sale securities	ASC 320-10-45-1	AFS			
Unrealized holdings gains and losses that result from a debt security being transferred into the available-for-sale category from the held-to-maturity category.	ASC 830-30-45-12	AFS			
Amounts recognized in other comprehensive income for debt securities classified as available-for-sale and held-to-maturity related to an other-than-temporary impairment recognized in accordance with Section 320-10-35 if a portion of the impairment was not recognized in earnings.	ASC 320-10-35	AFS			
Subsequent decreases (if not an other-than-temporary impairment) or increases in the fair value of available-for-sale securities previously written down as impaired.	ASC 320-10-35-18	AFS			
Gains and losses (effective portion) on derivative instruments that are designated as, and qualify as, cash flow hedges.	ASC 815-20-35-1(c)	DERIV			
Gains or losses associated with pension or other postretirement benefits (that are not recognized immediately as a component of net periodic benefit cost).	ASC 715-20-50-1(j)	PEN			
Prior service costs or credits associated with pension or other postretirement benefits.	ASC 715-20-50-1(j)	PEN			
Transition assets or obligations associated with pension or other postretirement benefits (that are not recognized immediately as a component of net periodic benefit cost).	ASC 715-20-50-1(j)	PEN			
Foreign currency translation adjustments	ASC 830-30-45-12	FC			
Gains and losses on foreign currency transactions that are designated as, and are effective as, economic hedges of a net investment in a foreign entity, commencing as of the designation date.	ASC 830-20-35-3(a)	FC			
Gains and losses on intra-entity foreign currency transactions that are of a long- term-investment nature (that is, settlement is not planned or anticipated in the foreseeable future), when the entities to the transaction are consolidated, combined, or accounted for by the equity method in the reporting entity's financial statements.	ASC 830-20-35-3(b)	FC			

Variable Name	Description	Source(s)
Bank Financial Characteristics		
AVGAT	Average total assets in millions	SNL Financial
GROSSLOANS	Total gross loans in millions, scaled by average total assets, x 100	SNL Financial
NETLOANS	Total net loans in millions, scaled by average total assets, x 100	SNL Financial
DEP	Total deposits in millions, scaled by average total assets, x 100	SNL Financial
BE	Total common equity in millions	SNL Financial
MVE	Year-end price multiplied by common shares outstanding (in	SNL Financial
	millions), supplemented with CRSP Monthly price and share data	CRSP MSF
NETINT	Net interest income in millions, scaled by average total assets, x	SNL Financial
LLP	Provision for loan losses in millions, scaled by average total	SNL Financial
DCL	assets, x 100	
RGL	Realized gains and losses on securities in millions, scaled by average total assets, then multiplied by 100	SNL Financial
CI	Comprehensive income in millions, scaled by average total assets, x 100. Missing values set equal to zero.	SNL Financial
NI	Net income in millions, scaled by average total assets, x 100. Missing values set equal to zero.	SNL Financial
OCI	Total other comprehensive income in millions, scaled by average total assets, x 100. Missing values set equal to zero.	SNL Financial
AFS	Available-for-sale securities adjustments in millions, scaled by average total assets, x 100. Missing values set equal to zero.	SNL Financial
AFSUGL	AES excluding AES recycled items. Hand-collected data obtained	SNL Financial
	from firm form 10-K and 10KSB on sec.gov.	SEC (sec.gov)
AFSRECY	AFS recycled items in millions, scaled by average total assets, x	SEC (sec.gov)
	100. Hand-collected data obtained from forms 10-K and 10KSB	~
	on sec.gov.	
DERIV	Cash-flow hedge adjustments in millions, scaled by average total	SNL Financial
	assets, x 100. Missing values set equal to zero.	
DERIVUGL	DERIV excluding AFS recycled items. Hand-collected data	SNL Financial
	obtained from firm form 10-K and 10KSB on sec.gov.	SEC (sec.gov)
DERIVRECY	DERIV recycled items in millions, scaled by average total assets,	SEC (sec.gov)
	x 100. Hand-collected data obtained from forms 10-K and	
	10KSB on sec.gov.	
PEN	Pension-related adjustments in millions, scaled by average total	SNL Financial
	assets, x 100. Missing values set equal to zero.	
FC	Foreign currency translation adjustments in millions, scaled by average total assets, x 100. Missing values set equal to zero.	SNL Financial
OTHER	Pref. dividend adjustments plus other items of OCI in millions.	SNL Financial
	scaled by average total assets, x 100. Missing values set equal to	
	zero.	
Returns Volatility		
σRET	Rolling five-year standard deviation of raw monthly stock returns	CRSP MSF
	(RET) x 100, requiring at least 20 monthly returns for each five-	
	vear period. Zero-return months excluded.	
Comprehensive Income Componen	t Volatilities and Control Variables	
σCI	Rolling five-year standard deviation of <i>CI</i>	SNL Financial
σNI	Rolling five-year standard deviation of <i>NI</i>	SNL Financial
σΟCΙ	Rolling five-year standard deviation of <i>OCI</i>	SNL Financial
σAFS	Rolling five-year standard deviation of AFS	SNL Financial
σDERIV	Rolling five-year standard deviation of DERIV	SNL Financial

Appendix B – Variable Definitions

σΡΕΝ	Rolling five-year standard deviation of <i>PEN</i>	SNL Financial
σFC	Rolling five-year standard deviation of <i>FC</i>	SNL Financial
σOTHER	Rolling five-year standard deviation of OTHER	SNL Financial
EXP	Total derivatives exposure, scaled by average total assets, x 100.	SNL Financial
	Missing values set equal to zero.	
GAP	One-year cumulative repricing gap, equal to the absolute value of	SNL Financial
	interest-sensitive assets minus interest-sensitive liabilities	
	schedules to reprice within one year in millions, scaled by	
	average total assets, x 100. Missing values set equal to zero.	
BTM	Book-to-market ratio (<i>BVE / MVE</i>)	SNL Financial
		CRSP MSF
PRICE_1	Lagged end-of-the year stock price, first from SNL Financial,	SNL Financial
	supplemented by CRSP MSF.	CRSP MSF
CI1	Lagged CI.	SNL Financial
ANNRET	Annual stock return from CRSP MSF. Four monthly returns	CRSP MSF
	required. Zero-return months excluded.	
ANNRET_1	Lagged ANNRET.	CRSP MSF
OCI Unrealized and Reali	ized Component Volatilities and Presentation Variables	
σAFSUGL	Rolling five-year standard deviation of the change in AFS	SEC (sec.gov)
	securities excluding recycled items in millions, scaled by average	
	total assets, x 100. Hand-collected data obtained from forms 10-	
	K and 10KSB on sec.gov.	
$\sigma AFSRECY$	Rolling five-year standard deviation of AFS recycled items in	SEC (sec.gov)
	millions, scaled by average total assets, x 100. Hand-collected	
	data obtained from forms 10-K and 10KSB on sec.gov.	
$\sigma DERIVUGL$	Rolling five-year standard deviation of the change in cash-flow	SEC (sec.gov)
	hedging instruments excluding recycled items in millions, scaled	
	by average total assets, x 100. Hand-collected data obtained from	
DEDUIDEQU	forms 10-K and 10KSB on sec.gov.	
$\sigma DERIVRECY$	Rolling five-year standard deviation of DERIV recycled items in	SEC (sec.gov)
	millions, scaled by average total assets, x 100. Hand-collected	
DDEG	data obtained from forms 10-K and 10KSB on sec.gov.	
PRES	Indicator variable equal to one if the firm presents OCI in a	SEC (sec.gov)
	performance statement consistent with ASC 220-10-45-1C; zero	
	otherwise. Hand-collected data obtained from forms 10-K and	
	10KSB on sec.gov.	

Table 1 – Sample Selection Criteria

Sample Requirements	Obs	Banks
Post-SFAS-130 (fiscal year beginning after December 15, 1997).	10,034	757
Observations from 2002-2012 from SNL Financial (observations with data from year t-4 to year t, post-SFAS-130).	7,828	754
Data for bank financial variables, CI components and their standard deviations, and primary control variables from SNL Financial.	6,962	728
Data for measure of five-year returns volatility (σRET) from the CRSP MSF. At least 20 returns required. FULL SAMPLE.	2,264	298
Data for CI, CI_1, ANNRET, ANNRET_1, and PRICE_1 (Performance Controls).	2,229	298
Hand-collected data for AFS recycling, DERIV recycling, and OCI presentation variables. RECYCLING SAMPLE.	898	121
Hand-collected data for AFS recycling, DERIV recycling, and OCI presentation variables and Performance Controls.	893	121

AVCAT 2.264 31,584,33 192,815,55 42,70 695,21 1,568,32 5,111,81 2,443,08,00 RCRSSLDANS 2.264 70,28 15,40 487 62,244 70,39 79,05 33,04 DEP 2.264 79,09 11,75 22,52 72,34 79,51 85,53 302,42 DWE 2.264 3,49,09 16,857,02 0.07 57,71 182,38 77,856 238,002,70 MVE 2.264 3,49,09 16,857,02 0.07 57,71 182,38 77,856 238,002,70 NETINT 2.264 0.60 0.31 -9,17 0.00 0.01 0.06 1.12 CI 2.264 0.60 1.13 -8,42 0.00 0.01 0.12 2.99 AFS 2.264 0.00 0.02 -3,49 -0.10 0.05 1.12 5.55 NI 2.264 0.00 0.01 -0.12 2.99 DERIV 2.264 0.00	VARIABLE	Ν	MEAN	STD	MIN	P25	P50	P75	MAX
GROSSLOAMS 2.264 70.28 15.40 4.87 66.18 67.44 77.83 336.30 DEF 2.264 60.12 15.28 485 61.68 67.44 77.83 336.30 DVE 2.264 2.79.63 15.990.17 -64.41 58.78 134.18 55.845 218.183.00 NETINT 2.264 3.490 0.73 0.90 2.99 3.36 3.76 7.955 LP 2.264 0.69 0.80 -1.22 0.15 0.31 0.73 7.10 RGL 2.264 0.60 1.13 -5.17 0.00 0.01 0.06 1.12 CI 2.264 0.60 1.13 -8.02 0.41 0.855 1.16 5.77 OCI 2.264 0.00 0.42 -3.37 -0.08 0.01 0.12 2.97 DERV 2.264 0.00 0.02 -0.33 0.00 0.00 0.00 0.02 0.24 3.30	AVGAT	2,264	31,584.33	192,815.55	42.70	695.21	1,568.32	5,111.81	2,443,068.00
NETLOANS 2.264 69.12 15.28 4.85 61.68 69.74 77.83 336.30 BVE 2.264 2.793.63 15.990.17 -64.41 58.78 134.18 528.45 218.188.00 MVE 2.264 3.490.09 16.857.02 0.07 57.71 182.38 77.86 238.023.70 NETINT 2.264 0.59 0.80 -1.22 0.15 0.31 0.73 7.10 RGL 2.264 0.60 1.13 -8.02 0.41 0.85 1.11 555 NI 2.264 0.60 1.13 -8.02 0.41 0.85 1.16 577 OCI 2.264 0.00 0.04 -0.45 0.00	GROSSLOANS	2,264	70.28	15.40	4.87	62.84	70.89	79.05	343.04
DEP 2.264 79.09 11.75 22.52 72.34 79.51 85.33 302.42 MVE 2.264 3.439.09 16.857.02 0.07 57.71 182.38 778.95 238.020.70 NETINT 2.264 0.59 0.80 -1.22 0.15 0.31 0.73 7.95 ILP 2.264 0.69 0.80 -1.22 0.15 0.31 0.73 7.10 RGL 2.264 0.60 1.13 -5.02 0.41 0.85 1.16 5.77 OCI 2.264 0.60 1.23 -3.49 -0.10 0.01 0.12 2.99 AFS 2.264 0.00 0.04 -0.67 0.00 0.00 0.00 0.00 0.02 2.99 AFS 2.264 0.00 0.02 -0.39 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00	NETLOANS	2,264	69.12	15.28	4.85	61.68	69.74	77.83	336.30
BVE 2,264 2,793,63 15,990,17 -64.41 58,78 134.18 528.45 218,188,00 MVFE 2,264 3,490,9 16,857,02 0.07 57,71 182.38 778,96 228,020,70 LLP 2,264 0.09 0.80 -1,22 0.15 0.31 0.73 7,10 RGL 2,264 0.00 0.31 -9,17 0.00 0.01 0.06 1.12 555 NI 2,264 0.00 0.29 -3,39 -0.10 0.01 0.12 2.97 AFS 2,264 0.00 0.04 -0.67 0.00 0.00 0.00 0.00 0.02 2.97 DERIV 2,264 0.00 0.04 -0.67 0.00 0.00 0.00 0.00 0.02 2.97 DERIV 2,264 0.00 0.04 -0.67 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00	DEP	2,264	79.09	11.75	22.52	72.34	79.51	85.53	302.42
NVE 2.264 3.439.09 16.857.02 0.07 57.71 182.38 778.96 228.002.70 NETINT 2.264 0.90 0.90 2.99 3.36 3.76 7.95 ILP 2.264 0.90 0.80 -1.22 0.15 0.31 0.73 7.10 RGL 2.264 0.00 1.17 -11.00 0.01 0.06 1.12 CI 2.264 0.00 1.13 -8.02 0.41 0.85 1.21 5.55 OCI 2.264 0.00 0.02 -3.49 -0.00 0.00 <td>BVE</td> <td>2,264</td> <td>2,793.63</td> <td>15,990.17</td> <td>-64.41</td> <td>58.78</td> <td>134.18</td> <td>528.45</td> <td>218,188.00</td>	BVE	2,264	2,793.63	15,990.17	-64.41	58.78	134.18	528.45	218,188.00
NETINT 2.264 3.40 0.73 0.90 2.99 3.36 3.76 7.95 LLP 2.264 0.05 0.80 -1.22 0.15 0.31 0.73 7.10 RGL 2.264 0.00 0.31 -9.17 0.00 0.01 0.06 1.12 CI 2.264 0.02 1.17 -11.20 0.39 0.85 1.16 5.55 OCI 2.264 0.01 0.29 -3.49 -0.10 0.01 0.12 2.97 DERIV 2.264 0.00 0.04 -0.67 0.00 0.00 0.00 0.00 0.02 2.97 DERIV 2.264 0.00 0.02 -0.39 0.00 0.0	MVE	2,264	3,439.09	16,857.02	0.07	57.71	182.38	778.96	238,020.70
LLP 2.64 0.59 0.80 -1.22 0.15 0.31 0.73 7.10 RGL 2.264 0.00 0.31 -9.17 0.00 0.01 0.06 1.12 CI 2.264 0.60 1.13 -8.02 0.41 0.85 1.21 5.55 NI 2.264 0.01 0.22 3.49 -0.10 0.01 0.12 2.97 DERIV 2.264 0.00 0.04 -0.45 0.00 </td <td>NETINT</td> <td>2.264</td> <td>3.40</td> <td>0.73</td> <td>0.90</td> <td>2.99</td> <td>3.36</td> <td>3.76</td> <td>7.95</td>	NETINT	2.264	3.40	0.73	0.90	2.99	3.36	3.76	7.95
RGL $2,264$ 0.00 0.31 -9.17 0.00 0.01 0.06 1.12 CI $2,264$ 0.60 1.13 -1.12 0.39 0.85 1.16 5.77 OCI $2,264$ 0.01 0.29 -3.34 -0.10 0.01 0.12 2.997 DERIV $2,264$ 0.00 0.04 -0.45 0.00 0.	LLP	2.264	0.59	0.80	-1.22	0.15	0.31	0.73	7.10
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	RGL	2.264	0.00	0.31	-9.17	0.00	0.01	0.06	1.12
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	CI	2.264	0.62	1.17	-11.20	0.39	0.85	1.21	5.55
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	NI	2,264	0.60	1.13	-8.02	0.41	0.85	1.16	5.77
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	OCI	2.264	0.01	0.29	-3.49	-0.10	0.01	0.12	2.99
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	AFS	2,264	0.02	0.27	-3.37	-0.08	0.01	0.12	2.97
PEN 2264 0.00 0.04 -0.67 0.00 0.00 0.00 0.02 FC 2.264 0.00 0.02 -0.39 0.00 0.00 0.00 0.22 FC 2.264 9.50 4.68 3.05 6.36 8.26 11.22 43.84 arCl 2.264 0.55 0.76 0.01 0.14 0.22 6.63 13.14 arCl 2.264 0.55 0.76 0.01 0.14 0.26 0.63 13.14 arCl 2.264 0.22 0.22 0.00 0.07 0.16 2.07 arCl 2.264 0.22 0.22 0.00 0.01 0.17 0.27 2.30 arAs 2.264 0.02 0.04 0.00 0.00 0.00 0.00 0.00 arAs 2.264 0.02 0.02 0.00 0.00 0.00 0.00 0.00 arAs 2.264 0.02 0.05	DERIV	2,264	0.00	0.04	-0.45	0.00	0.00	0.00	0.69
FC 2,264 0.00 0.02 -0.39 0.00 <t< td=""><td>PEN</td><td>2,264</td><td>0.00</td><td>0.04</td><td>-0.67</td><td>0.00</td><td>0.00</td><td>0.00</td><td>0.22</td></t<>	PEN	2,264	0.00	0.04	-0.67	0.00	0.00	0.00	0.22
OTHER 2.264 -0.01 0.06 -0.63 0.00 0.00 0.00 0.49 dRET 2.264 9.50 4.68 3.05 6.36 8.26 11.22 43.84 σCI 2.264 0.55 0.76 0.01 0.14 0.26 0.63 13.14 σCI 2.264 0.15 0.76 0.01 0.14 0.26 0.63 13.14 σCI 2.264 0.22 0.22 0.00 0.17 0.27 2.35 aFS 2.264 0.22 0.22 0.00 0.10 0.17 0.27 2.35 aFE 2.264 0.02 0.04 0.00	FC	2,264	0.00	0.02	-0.39	0.00	0.00	0.00	0.26
σRET 2.264 9.50 4.68 3.05 6.36 8.26 11.22 43.84 σCI 2.264 0.65 0.77 0.04 0.24 0.39 0.74 13.00 σCI 2.264 0.55 0.76 0.01 0.14 0.26 0.63 13.14 σCI 2.264 0.24 0.23 0.00 0.07 0.16 2.07 σCI 2.264 0.22 0.22 0.00 0.10 0.17 0.27 2.33 σFE 2.264 0.02 0.04 0.00	OTHER	2,264	-0.01	0.06	-0.63	0.00	0.00	0.00	0.49
σCI 2.264 0.65 0.77 0.04 0.24 0.39 0.74 1.300 σNI 2.264 0.55 0.76 0.01 0.14 0.26 0.63 13.14 σCI 2.264 0.22 0.22 0.00 0.12 0.19 0.29 2.35 σAFS 2.264 0.22 0.22 0.00 0.11 0.27 2.30 $\sigma BERN$ 2.264 0.02 0.04 0.00	σRET	2,264	9 50	4 68	3.05	636	8.26	11.22	43.84
ord 2264 0.55 0.76 0.01 0.14 0.26 0.63 1.14 aCI-aVI 2264 0.10 0.18 -0.65 0.00 0.07 0.16 2.07 aCI 2.264 0.22 0.22 0.00 0.12 0.19 0.29 2.35 aLFS 2.264 0.02 0.04 0.00 0.00 0.00 0.01 0.17 0.27 2.30 aDERIV 2.264 0.02 0.04 0.00 0.02 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.0	σCI	2,264	0.65	0.77	0.04	0.24	0.20	0.74	13.00
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	σNI	2,261	0.55	0.76	0.01	0.14	0.26	0.63	13.00
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	σCI - σΝΙ	2,264	0.55	0.18	-0.65	0.00	0.20	0.05	2 07
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	σΟCΙ	2,264	0.10	0.10	0.00	0.00	0.19	0.10	2.07
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	σAFS	2,204	0.24	0.23	0.00	0.12	0.17	0.27	2.35
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	σDFRIV	2,204	0.02	0.22	0.00	0.10	0.00	0.01	0.38
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	σPEN	2,204	0.02	0.04	0.00	0.00	0.00	0.01	0.30
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	σEC	2,204	0.01	0.03	0.00	0.00	0.00	0.00	0.30
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	of C other	2,204	0.00	0.02	0.00	0.00	0.00	0.00	0.20
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	FYP	2,204	46.80	336.05	0.00	0.00	0.00	5.55	5 285 21
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	GAP	2,204	40.00 6.97	10.16	0.00	0.00	1.00	10.00	52.04
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	BTM	2,204	1.03	1 00	-19.34	0.00	0.80	1 10	14.96
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	DIM PRICE 1	2,204	27.56	50.08	-17.54	0.55	17.34	27.81	1 368 80
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	CL1	2,204	0.63	1 24	-15.41	0.40	0.86	1.24	5 55
AINMELT2,2462,7435.5255.74 -10.53 22.5522.5522.5022.55 AFS / OCI 2,2460.858.86 -311.62 0.851.001.00222.11 $DERIV / OCI$ 2,2460.157.42 -84.50 0.000.000.00325.44 PEN / OCI 2,246-0.021.21 -44.29 0.000.000.0010.13 FC / OCI 2,246-0.115.78 -214.07 0.000.000.0012.11 $ANNRET$ 2,2324.4136.05 -93.47 -16.09 3.9823.09313.03 $AFSUGL$ 8980.020.30 -3.06 -0.08 0.020.131.72 $AFSRECY$ 8980.000.18 -0.57 -0.02 0.000.000.00 $DERIVUGL$ 8980.000.03 -0.46 0.000.000.000.30 $\sigma AFSUGL$ 8980.020.050.000.010.030.881.16 $\sigma ERIVUGL$ 8980.020.050.000.000.000.25 $PRES$ 8980.020.050.000.000.000.25 $PERIVUGL / OCI8970.3711.22-308.110.480.981.14\sigma AFSECY / OCI8970.3711.22-308.110.480.981.14\sigma AFSUGL / OCI8970.3711.22-308.110.480.981.1450.91$	ANNRET 1	2,200	2.04	35.02	-03.47	-16.95	2.58	22.09	275.36
AFS/OCI2,2400.056.00-11020.051.001.00222.11 $DERIV/OCI$ 2,2460.157.42-84.500.000.000.00325.44 PEN/OCI 2,246-0.021.21-44.290.000.000.0010.13 FC/OCI 2,2460.010.45-7.200.000.000.0017.71 $OTHER/OCI$ 2,246-0.115.78-214.070.000.000.0012.11 $ANNRET$ 2,2324.4136.05-93.47-16.093.9823.09313.03 $AFSUGL$ 8980.020.30-3.06-0.080.020.131.72 $AFSRECY$ 8980.000.06-0.510.000.000.002.55 $DERIVUGL$ 8980.000.06-0.510.000.000.000.30 $\sigma AFSUGL$ 8980.070.140.000.010.030.081.16 $\sigma DERIVECY$ 8980.020.050.000.000.000.000.25 $PRES$ 8980.020.050.000.000.000.000.25 $PRES$ 8980.300.460.000.000.000.25 $PRES$ 8980.300.460.000.000.000.25 $PRES$ 8980.3711.22-308.110.480.981.1450.91 $AFSUGL/OCI$ 8970.3711.22-308.1	ANNKEI_I AFS / OCI	2,240	2.94	8 86	-75.47	-10.95	2.58	1.00	275.50
DERIV / OC12,2460.157.42-84.500.000.000.000.00525.44PEN / OC12,246-0.021.21-44.290.000.000.0010.13FC / OC12,2460.010.45-7.200.000.000.0017.71OTHER / OC12,246-0.115.78-214.070.000.000.0012.11ANNRET2,2324.4136.05-93.47-16.093.9823.09313.03AFSUGL8980.020.30-3.06-0.080.020.131.72AFSRECY8980.000.18-0.57-0.020.000.002.55DERIVUGL8980.000.03-0.460.000.000.000.30 $\sigma AFSUGL$ 8980.240.230.000.110.180.301.83 $\sigma AFSRECY$ 8980.070.140.000.010.030.081.16 $\sigma DERIVUGL$ 8980.020.050.000.000.000.010.35 $\sigma AFSRECY$ 8980.010.020.000.000.000.010.35 $\sigma AFSRECY$ 8980.010.020.000.000.000.000.25DERIVUGL8980.020.050.000.000.000.000.25PRES8980.300.460.000.000.000.000.25PRES8970.37<		2,240	0.85	8.80 7.40	-511.02	0.85	1.00	1.00	222.11
PEN/OCI $2,246$ -0.02 1.21 -44.29 0.00 0.00 0.00 10.13 FC/OCI $2,246$ 0.01 0.45 -7.20 0.00 0.00 0.00 10.13 OTHER/OCI $2,246$ -0.11 5.78 -214.07 0.00 0.00 0.00 12.11 ANNRET $2,232$ 4.41 36.05 -93.47 -16.09 3.98 23.09 313.03 AFSUGL 898 0.02 0.30 -3.06 -0.08 0.02 0.13 1.72 AFSRECY 898 0.00 0.18 -0.57 -0.02 0.00 0.00 2.55 DERIVUGL 898 0.00 0.06 -0.51 0.00 0.00 0.00 0.69 DERIVECY 898 0.00 0.03 -0.46 0.00 0.00 0.00 0.30 $\sigma AFSUGL$ 898 0.02 0.05 0.00 0.01 0.03 0.08 1.16 $\sigma DERIVUGL$ 898 0.02 0.05 0.00 0.00 0.00 0.00 0.00 0.02 DERIVUGL 898 0.01 0.02 0.00 0.00 0.00 0.00 0.00 0.00 $\sigma AFSRECY$ 898 0.01 0.02 0.00 0.00 0.00 0.00 0.00 0.00 $\sigma AFSRECY$ 898 0.01 0.02 0.00 0.00 0.00 0.00 0.00 0.00 $\sigma DERIVRECY$ 898 0.01 <td>DERIV / OCI</td> <td>2,240</td> <td>0.15</td> <td>7.42</td> <td>-84.50</td> <td>0.00</td> <td>0.00</td> <td>0.00</td> <td>325.44</td>	DERIV / OCI	2,240	0.15	7.42	-84.50	0.00	0.00	0.00	325.44
FC / OCI2,2460.010.45 -7.20 0.000.000.0017.71OTHER / OCI2,246 -0.11 5.78 -214.07 0.000.000.0012.11ANNRET2,2324.4136.05 -93.47 -16.09 3.9823.09313.03AFSUGL8980.020.30 -3.06 -0.08 0.020.131.72AFSRECY8980.000.18 -0.57 -0.02 0.000.000.002.55DERIVUGL8980.000.03 -0.46 0.000.000.000.30 $\sigma AFSUGL$ 8980.240.230.000.110.180.301.83 $\sigma AFSUGL$ 8980.020.050.000.010.030.081.16 $\sigma DERIVUGL$ 8980.020.050.000.000.000.000.25PRES8980.010.020.000.000.000.000.25PRES8980.300.460.000.000.000.000.25PRES8980.300.460.000.000.000.000.25PRES8980.300.460.000.000.000.0953.84DERIVUGL/OCI8970.163.56-24.23-0.100.000.001.367.84DERIVUGL/OCI8971.5845.73-48.560.000.000.001.367.84	PEN/OCI	2,246	-0.02	1.21	-44.29	0.00	0.00	0.00	10.13
OTHER/OCI2,246-0.115.78-214.070.000.000.0012.11ANNRET2,2324.4136.05-93.47-16.093.9823.09313.03AFSUGL8980.020.30-3.06-0.080.020.131.72AFSRECY8980.000.18-0.57-0.020.000.002.55DERIVUGL8980.000.06-0.510.000.000.000.69DERIVRECY8980.000.03-0.460.000.000.000.30 $\sigma AFSUGL$ 8980.240.230.000.110.180.301.83 $\sigma AFSRECY$ 8980.070.140.000.010.030.081.16 $\sigma DERIVUGL$ 8980.020.050.000.000.000.010.35 $\sigma DERIVRECY$ 8980.010.020.000.000.000.02PRES8980.300.460.000.000.000.000.25PRES8970.3711.22-308.110.480.981.1450.91AFSRECY/OCI8970.163.56-24.23-0.100.000.001.367.84DERIVUGL/OCI8971.5845.73-48.560.000.000.001.367.84	FC/OCI	2,246	0.01	0.45	-7.20	0.00	0.00	0.00	1/./1
ANNRET2,2324.4136.05 -93.47 -16.09 3.98 23.09 313.03 AFSUGL8980.020.30 -3.06 -0.08 0.020.13 1.72 AFSRECY8980.000.18 -0.57 -0.02 0.000.002.55DERIVUGL8980.000.06 -0.51 0.000.000.000.69DERIVRECY8980.000.03 -0.46 0.000.000.000.30 $\sigma AFSUGL$ 8980.240.230.000.110.180.301.83 $\sigma AFSRECY$ 8980.070.140.000.010.030.081.16 $\sigma DERIVUGL$ 8980.020.050.000.000.000.010.35 $\sigma DERIVRECY$ 8980.010.020.000.000.000.000.25PRES8980.300.460.000.000.000.000.25PRES8980.300.460.000.000.000.000.25PRES8970.3711.22 -308.11 0.480.981.1450.91AFSRECY/OCI8970.163.56 -24.23 -0.10 0.000.001.367.84DERIVRECY/OCI8971.5845.73 -48.56 0.000.000.001.367.84	OTHER / OCI	2,246	-0.11	5.78	-214.07	0.00	0.00	0.00	12.11
AFSUGL898 0.02 0.30 -3.06 -0.08 0.02 0.13 1.72 AFSRECY898 0.00 0.18 -0.57 -0.02 0.00 0.00 2.55 DERIVUGL898 0.00 0.06 -0.51 0.00 0.00 0.00 0.69 DERIVRECY898 0.00 0.03 -0.46 0.00 0.00 0.00 0.30 $\sigma AFSUGL$ 898 0.24 0.23 0.00 0.11 0.18 0.30 1.83 $\sigma AFSRECY$ 898 0.07 0.14 0.00 0.01 0.03 0.08 1.16 $\sigma DERIVUGL$ 898 0.02 0.05 0.00 0.00 0.00 0.01 0.35 $\sigma DERIVRECY$ 898 0.01 0.02 0.00 0.00 0.00 0.00 0.01 0.35 $\sigma DERIVRECY$ 898 0.01 0.02 0.00 0.00 0.00 0.00 0.00 0.00 $\sigma DERIVRECY$ 898 0.30 0.46 0.00 0.00 0.00 0.00 0.00 $AFSUGL/OCI$ 897 0.37 11.22 -308.11 0.48 0.98 1.14 50.91 $AFSRECY/OCI$ 897 0.16 3.56 -24.23 -0.10 0.00 0.00 $1.367.84$ $DERIVRECY/OCI$ 897 1.58 45.73 -48.56 0.00 0.00 0.00 $1.367.84$	ANNRET	2,232	4.41	36.05	-93.47	-16.09	3.98	23.09	313.03
AFSRECY8980.000.18 -0.57 -0.02 0.000.002.55DERIVUGL8980.000.06 -0.51 0.000.000.000.69DERIVRECY8980.000.03 -0.46 0.000.000.000.30 $\sigma AFSUGL$ 8980.240.230.000.110.180.301.83 $\sigma AFSRECY$ 8980.070.140.000.010.030.081.16 $\sigma DERIVUGL$ 8980.020.050.000.000.000.010.35 $\sigma DERIVRECY$ 8980.010.020.000.000.000.000.25PRES8980.300.460.000.000.001.001.00AFSUGL/OCI8970.3711.22-308.110.480.981.1450.91AFSRECY/OCI8970.163.56-24.23-0.100.000.001.367.84DERIVRECY/OCI8971.5845.73-48.560.000.000.003.42	AFSUGL	898	0.02	0.30	-3.06	-0.08	0.02	0.13	1.72
DERIVUGL8980.000.06 -0.51 0.000.000.000.69DERIVRECY8980.000.03 -0.46 0.000.000.000.30 $\sigma AFSUGL$ 8980.240.230.000.110.180.301.83 $\sigma AFSRECY$ 8980.070.140.000.010.030.081.16 $\sigma DERIVUGL$ 8980.020.050.000.000.000.010.35 $\sigma DERIVRECY$ 8980.010.020.000.000.000.000.25PRES8980.300.460.000.000.001.001.00AFSUGL/OCI8970.3711.22 -308.11 0.480.981.1450.91AFSRECY/OCI8970.163.56 -24.23 -0.10 0.000.001.367.84DERIVRECY/OCI8971.5845.73 -48.56 0.000.000.003.42	AFSRECY	898	0.00	0.18	-0.57	-0.02	0.00	0.00	2.55
DERIVRECY8980.000.03-0.460.000.000.000.30 $\sigma AFSUGL$ 8980.240.230.000.110.180.301.83 $\sigma AFSRECY$ 8980.070.140.000.010.030.081.16 $\sigma DERIVUGL$ 8980.020.050.000.000.000.010.35 $\sigma DERIVRECY$ 8980.010.020.000.000.000.000.000.25PRES8980.300.460.000.000.001.001.00AFSUGL/OCI8970.3711.22-308.110.480.981.1450.91AFSRECY/OCI8970.163.56-24.23-0.100.000.0953.84DERIVRECY/OCI8971.5845.73-48.560.000.000.001.367.84DERIVRECY/OCI897-0.9928.96-867.160.000.000.003.42	DERIVUGL	898	0.00	0.06	-0.51	0.00	0.00	0.00	0.69
$\sigma AFSUGL$ 8980.240.230.000.110.180.301.83 $\sigma AFSRECY$ 8980.070.140.000.010.030.081.16 $\sigma DERIVUGL$ 8980.020.050.000.000.000.010.35 $\sigma DERIVRECY$ 8980.010.020.000.000.000.000.000.25PRES8980.300.460.000.000.001.001.00AFSUGL/OCI8970.3711.22-308.110.480.981.1450.91AFSRECY/OCI8970.163.56-24.23-0.100.000.0953.84DERIVUGL/OCI8971.5845.73-48.560.000.000.003.42	DERIVRECY	898	0.00	0.03	-0.46	0.00	0.00	0.00	0.30
$\sigma AFSRECY$ 8980.070.140.000.010.030.081.16 $\sigma DERIVUGL$ 8980.020.050.000.000.000.010.35 $\sigma DERIVRECY$ 8980.010.020.000.000.000.000.000.25PRES8980.300.460.000.000.001.001.00AFSUGL/OCI8970.3711.22-308.110.480.981.1450.91AFSRECY/OCI8970.163.56-24.23-0.100.000.0953.84DERIVUGL/OCI8971.5845.73-48.560.000.000.001.367.84DERIVRECY/OCI897-0.9928.96-867.160.000.000.003.42	σAFSUGL	898	0.24	0.23	0.00	0.11	0.18	0.30	1.83
$\sigma DERIVUGL$ 898 0.02 0.05 0.00 0.00 0.00 0.01 0.35 $\sigma DERIVRECY$ 898 0.01 0.02 0.00 0.00 0.00 0.00 0.00 0.00 0.25 PRES 898 0.30 0.46 0.00 0.00 0.00 1.00 1.00 AFSUGL/OCI 897 0.37 11.22 -308.11 0.48 0.98 1.14 50.91 AFSRECY/OCI 897 0.16 3.56 -24.23 -0.10 0.00 0.09 53.84 DERIVUGL/OCI 897 1.58 45.73 -48.56 0.00 0.00 0.00 1,367.84 DERIVRECY/OCI 897 -0.99 28.96 -867.16 0.00 0.00 0.00 3.42	σAFSRECY	898	0.07	0.14	0.00	0.01	0.03	0.08	1.16
$\sigma DERIVRECY$ 898 0.01 0.02 0.00 0.00 0.00 0.00 0.25 PRES 898 0.30 0.46 0.00 0.00 0.00 1.00 1.00 AFSUGL/OCI 897 0.37 11.22 -308.11 0.48 0.98 1.14 50.91 AFSRECY/OCI 897 0.16 3.56 -24.23 -0.10 0.00 0.09 53.84 DERIVUGL/OCI 897 1.58 45.73 -48.56 0.00 0.00 0.00 1,367.84 DERIVRECY/OCI 897 -0.99 28.96 -867.16 0.00 0.00 0.00 3.42	$\sigma DERIVUGL$	898	0.02	0.05	0.00	0.00	0.00	0.01	0.35
PRES 898 0.30 0.46 0.00 0.00 0.00 1.00 1.00 AFSUGL/OCI 897 0.37 11.22 -308.11 0.48 0.98 1.14 50.91 AFSRECY/OCI 897 0.16 3.56 -24.23 -0.10 0.00 0.09 53.84 DERIVUGL/OCI 897 1.58 45.73 -48.56 0.00 0.00 0.00 1,367.84 DERIVRECY/OCI 897 -0.99 28.96 -867.16 0.00 0.00 0.00 3.42	$\sigma DERIVRECY$	898	0.01	0.02	0.00	0.00	0.00	0.00	0.25
AFSUGL/OCI 897 0.37 11.22 -308.11 0.48 0.98 1.14 50.91 AFSRECY/OCI 897 0.16 3.56 -24.23 -0.10 0.00 0.09 53.84 DERIVUGL/OCI 897 1.58 45.73 -48.56 0.00 0.00 0.00 1,367.84 DERIVRECY/OCI 897 -0.99 28.96 -867.16 0.00 0.00 0.00 3.42	PRES	898	0.30	0.46	0.00	0.00	0.00	1.00	1.00
AFSRECY/OCI 897 0.16 3.56 -24.23 -0.10 0.00 0.09 53.84 DERIVUGL/OCI 897 1.58 45.73 -48.56 0.00 0.00 0.00 1,367.84 DERIVRECY/OCI 897 -0.99 28.96 -867.16 0.00 0.00 3.42	AFSUGL / OCI	897	0.37	11.22	-308.11	0.48	0.98	1.14	50.91
DERIVUGL/OCI 897 1.58 45.73 -48.56 0.00 0.00 1,367.84 DERIVRECY/OCI 897 -0.99 28.96 -867.16 0.00 0.00 3.42	AFSRECY/OCI	897	0.16	3.56	-24.23	-0.10	0.00	0.09	53.84
DERIVRECY/OCI 897 -0.99 28.96 -867.16 0.00 0.00 0.00 3.42	DERIVUGL / OCI	897	1.58	45,73	-48.56	0.00	0.00	0.00	1.367.84
	DERIVRECY / OCI	897	-0.99	28.96	-867.16	0.00	0.00	0.00	3.42

Table 2 – Descriptive Statistics

	Τa	ıbl	le	3 –	• C	Corre	elati	ion	St	ati	isti	ics	for	P	rimar	v F	Rea	gres	ssior	ı١	/ari	ab	les
--	----	-----	----	-----	-----	-------	-------	-----	----	-----	------	-----	-----	---	-------	-----	-----	------	-------	----	------	----	-----

Note: Pearson (above the diagonal) and Spearman (below the diagonal) correlation statistics. * indicates statistical significance at the 5% level based on twotailed tests.

VARIABLE	σRET	σCI	σNI	σCI - σNI	¢OCI	oAFS	øDERIV	σPEN	øFC	σOTHER	EXP	GAP	oAFSUGL	oAFSRECY	øDERIVUGL	øDERIVRECY	PRES
σRET		0.50*	0.52*	-0.07*	0.04*	0.04	0.08*	-0.09*	0.04*	0.06*	0.02	0.07*	0.03	0.26*	0.02	0.07*	0.16*
σCI	0.56*		0.97*	0.15*	0.26*	0.26*	0.08*	-0.00	0.06*	0.03	-0.01	0.04*	0.39*	0.35*	0.04	0.02	0.06
σNI	0.60*	0.82*		-0.08*	0.10*	0.10*	0.06*	-0.02	0.03	-0.00	-0.02	0.04*	0.20*	0.31*	0.02	0.02	0.09*
σCI - σNI	-0.14*	0.15*	-0.34*		0.70*	0.70*	0.09*	0.06*	0.13*	0.12*	0.01	-0.00	0.70*	0.20*	0.07*	0.02	-0.09*
σΟCΙ	-0.03	0.28*	-0.06*	0.60*		0.97*	0.12*	0.09*	0.25*	0.19*	0.05*	-0.04	0.91*	0.56*	0.08*	0.05	-0.09*
σAFS	-0.06*	0.23*	-0.09*	0.57*	0.91*		0.04*	0.05*	0.18*	0.08*	0.02	-0.02	0.94*	0.56*	-0.01	-0.04	-0.07*
$\sigma DERIV$	0.09*	0.14*	0.11*	0.09*	0.18*	0.08*		0.08*	0.16*	0.00	0.15*	-0.07*	-0.01	-0.05	0.97*	0.61*	-0.06
σPEN	-0.18*	-0.03	-0.06*	0.09*	0.14*	0.09*	0.15*		0.03	-0.06*	-0.02	-0.05*	0.02	-0.04	0.03	0.01	-0.01
σFC	0.05*	0.07*	0.08*	0.06*	0.09*	0.01	0.35*	0.19*		0.06*	0.20*	-0.07*	0.11*	0.07*	0.19*	0.32*	-0.08*
$\sigma OTHER$	0.12*	0.13*	0.13*	0.07*	0.14*	0.01	0.17*	0.00	0.12*		0.03	-0.05*	0.04	0.19*	-0.03	-0.01	-0.01
EXP	0.11*	0.10*	0.11*	0.01	0.12*	0.02	0.59*	0.21*	0.40*	0.18*		-0.09*	0.06	0.01	0.22*	0.03	-0.07*
GAP	0.06*	-0.03	-0.03	-0.03	-0.06*	-0.03	-0.10*	-0.07*	-0.16*	-0.13*	-0.14*		0.04	0.00	0.01	-0.06	0.06
$\sigma AFSUGL$	-0.07*	0.24*	-0.10*	0.62*	0.86*	0.95*	0.08*	0.09*	0.05	-0.06	0.06	-0.05		0.65*	-0.02	-0.06	-0.08*
$\sigma AFSRECY$	0.12*	0.24*	0.24*	0.03	0.29*	0.36*	0.03	0.06	0.01	0.05	0.13*	-0.08*	0.40*		-0.06	-0.03	-0.01
$\sigma DERIVUGL$	0.03	0.15*	0.07*	0.14*	0.21*	0.09*	0.96*	0.10*	0.33*	0.18*	0.54*	-0.04	0.08*	0.03		0.66*	-0.04
$\sigma DERIVRECY$	0.10*	0.17*	0.11*	0.06	0.17*	0.09*	0.59*	0.11*	0.31*	0.16*	0.40*	-0.08*	0.07*	0.13*	0.62*		-0.05
PRES	0.18*	0.02	0.07*	-0.11*	-0.17*	-0.13*	-0.06	-0.04	-0.13*	-0.04	-0.02	0.07*	-0.12*	0.04	-0.01	0.04	

Table 4 – Returns Volatility, Comprehensive Income Volatility, and OCI Volatility

$$\sigma RET_{it} = \beta_0 + \beta_1 \sigma CI \ COMPONENT_{it} + \beta_2 EXP_{it} + \beta_3 GAP_{it} + \sum_{10} \gamma_{2002-2011} + \varepsilon_{it}$$
(1)

$$\sigma RET_{it} = \beta_0 + \beta_1 \sigma NI_{it} + \beta_2 \sigma OCIProxy_{it} + \beta_3 EXP_{it} + \beta_4 GAP_{it} + \sum_{10} \gamma_{2002-2011} + \varepsilon_{it}$$
(2)

Note: The table presents the results of estimating equations (1) and (2) using the full sample. $\sigma CI COMPONENT_{it}$ equals σNI_{it} or σCI_{it} . $\sigma OCIProxy_{it}$ equals $\sigma CI_{it} - \sigma NI_{it}$ or σOCI_{it} . Variables are defined in Appendix B. All estimations include year fixed effects and a constant term, which are not presented for the sake of brevity. Regression coefficients are presented above t-statistics calculated based on standard errors clustered at the firm level. ***, **, and * indicate statistical significance at the 0.01, 0.05, and 0.10 levels based on two-tailed tests.

Dependent Variable =		σR	ET	
Model =	Equat	ion (1)	Equat	ion (2)
PARAMETER	(1)	(2)	(3)	(4)
σNI	2.40***		2.38***	2.42***
	(4.94)		(4.92)	(5.00)
σCI		2.23***		
		(4.55)		
σCI - σΝΙ			-1.20	
			(-1.10)	
σΟCΙ				-0.85
				(-0.94)
EXP	0.00**	0.00**	0.00*	0.00**
	(2.06)	(2.39)	(1.92)	(1.98)
GAP	0.03*	0.03*	0.03*	0.03*
	(1.79)	(1.81)	(1.78)	(1.74)
YEAR FE	YES	YES	YES	YES
N	2,264	2,264	2,264	2,264
R ²	0.43	0.42	0.43	0.43

Table 5 – Returns Volatility and OCI Component Volatilities

$$\sigma RET_{it} = \beta_0 + \beta_1 \sigma AFS_{it} + \beta_2 \sigma DERIV_{it} + \beta_3 \sigma PEN_{it} + \beta_4 \sigma FC_{it} + \beta_5 \sigma OTHER_{it} + \beta_6 \sigma NI_{it} + \sum_j \delta_j CONTROLS_{it} + \sum_{10} \gamma_{2002-2011} + \varepsilon_{it}$$
(3)

Note: The table presents the results of estimating equation (3). Columns (1)-(3) present results using the full sample. Columns (4)-(6) present results using the recycling sample. In Columns (1) and (4), $CONTROLS_{it}$ includes derivatives exposure (EXP_{it}) and interest rate gap (GAP_{it}) . In Columns (2) and (5), $CONTROLS_{it}$ includes derivatives exposure (EXP_{it}) interest rate gap (GAP_{it}) , firm size (market value of equity - MVE_{it}), and book-to-market ratio (BTM_{it}) . In Columns (3) and (6), $CONTROLS_{it}$ includes derivatives exposure (EXP_{it}) interest rate gap (GAP_{it}) , firm size (market value of equity - MVE_{it}) interest rate gap (GAP_{it}) , firm size (market value of equity - MVE_{it}) interest rate gap (GAP_{it}) , firm size (market value of equity - MVE_{it}), book-to-market ratio (BTM_{it}) , comprehensive income (CI_{it-1}) , annual stock return $(ANNRET_{it})$, lagged annual stock return $(ANNRET_{it-1})$ and lagged price $(PRICE_{it-1})$. All estimations include year fixed effects, and a constant term. Control variables, year fixed effects, and the constant term are not presented for the sake of brevity. Regression coefficients are presented above t-statistics calculated based on standard errors clustered at the firm level. ***, **, and * indicate statistical significance at the 0.01, 0.05, and 0.10 levels based on two-tailed tests.

Dependent Variable =	σRET										
Model =			Equati	on (3)							
PARAMETER	(1)	(2)	(3)	(4)	(5)	(6)					
σAFS	-0.60	-0.57	-0.22	-2.29*	-2.26*	-1.53					
	(-0.61)	(-0.60)	(-0.28)	(-1.69)	(-1.70)	(-1.29)					
σDERIV	1.33	1.66	1.28	-3.58	-2.64	-1.61					
	(0.36)	(0.46)	(0.43)	(-0.90)	(-0.69)	(-0.47)					
σPEN	-14.21***	-13.81***	-12.92***	-14.81**	-14.81***	-14.15***					
	(-4.21)	(-4.25)	(-4.46)	(-2.58)	(-2.63)	(-2.85)					
σFC	9.31*	10.29**	13.44**	11.03**	12.01**	15.10**					
	(1.81)	(2.11)	(2.40)	(2.21)	(2.27)	(2.54)					
σOTHER	-5.17	-4.48	-3.09	-3.69	-3.63	-2.20					
	(-1.48)	(-1.29)	(-0.92)	(-0.63)	(-0.62)	(-0.37)					
σΝΙ	2.35***	2.31***	1.50***	2.92***	2.89***	1.91***					
	(4.90)	(4.92)	(3.58)	(6.56)	(6.58)	(5.02)					
EXP and GAP	YES	YES	YES	YES	YES	YES					
YEAR FE	YES	YES	YES	YES	YES	YES					
MVE and BTM	NO	YES	YES	NO	YES	YES					
PERFORMANCE CONTROLS	NO	NO	YES	NO	NO	YES					
(H2) F-stat ($\beta_1 = \beta_2 = \beta_3 = \beta_4 = \beta_5$)	5.06***	5.42***	6.03***	3.54***	3.61***	4.26***					
(H2) F-stat ($\beta_1 = \beta_2 = \beta_3 = \beta_4 = \beta_5 = 0$)	4.28***	4.51***	5.08***	3.41***	3.52***	4.36***					
Ν	2,264	2,264	2,229	898	898	893					
<u>R²</u>	0.45	0.45	0.51	0.44	0.45	0.49					

Table 6 - Returns Volatility and OCI Unrealized and Recycled Subcomponent Volatilities

$$\sigma RET_{it} = \beta_0 + \beta_1 \sigma AFSUGL_{it} + \beta_2 \sigma AFSRECY_{it} + \beta_3 \sigma DERIVUGL_{it} + \beta_4 \sigma DERIVRECY_{it} + \beta_5 \sigma PEN_{it} + \beta_6 FC_{it} + \beta_7 \sigma OTHER_{it} + \beta_8 \sigma NI_{it} + \sum_j \delta_j CONTROLS_{it} + \sum_{10} \gamma_{2002-2011} + \varepsilon_{it}$$
(4)

Note: The table presents the results of estimating equation (4) using the recycling sample. In Column (1), $CONTROLS_{it}$ includes derivatives exposure (EXP_{it}) and interest rate gap (GAP_{it}) . In Column (2), $CONTROLS_{it}$ includes derivatives exposure (EXP_{it}) interest rate gap (GAP_{it}) , firm size (market value of equity - MVE_{it}), and book-to-market ratio (BTM_{it}) . In Column (3), $CONTROLS_{it}$ includes derivatives exposure (EXP_{it}) interest rate gap (GAP_{it}) , firm size (market value of equity - MVE_{it}), and book-to-market ratio (BTM_{it}) . In Column (3), $CONTROLS_{it}$ includes derivatives exposure (EXP_{it}) interest rate gap (GAP_{it}) , firm size (market value of equity - MVE_{it}), book-to-market ratio (BTM_{it}) , comprehensive income (CI_{it}) , lagged comprehensive income (CI_{it-1}) , annual stock return $(ANNRET_{it})$, lagged annual stock return $(ANNRET_{it-1})$ and lagged price $(PRICE_{it-1})$. All estimations include year fixed effects, and a constant term. Control variables, year fixed effects, and the constant term are not presented for the sake of brevity. Regression coefficients are presented above t-statistics calculated based on standard errors clustered at the firm level. ***, **, and * indicate statistical significance at the 0.01, 0.05, and 0.10 levels based on two-tailed tests.

Dependent Variable =	σRET						
Model =]	Equation (4)				
PARAMETER	(1)	(2)	(3)				
σAFSUGL	-5.13***	-5.06***	-3.87***				
	(-3.15)	(-3.24)	(-2.90)				
σAFSRECY	7.62*	7.55*	6.69*				
	(1.86)	(1.87)	(1.71)				
σDERIVUGL	-9.62**	-8.38**	-7.19**				
	(-2.27)	(-2.20)	(-2.14)				
σDERIVRECY	20.55	19.47	19.64				
	(1.59)	(1.56)	(1.65)				
σPEN	-12.89***	-12.88***	-12.42***				
	(-2.65)	(-2.69)	(-2.93)				
σFC	6.19	7.31	10.27				
	(0.79)	(0.91)	(1.28)				
σΟΤΗΕR	-5.76	-5.69	-4.22				
	(-1.19)	(-1.18)	(-0.89)				
σNI	2.62***	2.59***	1.66***				
	(6.56)	(6.54)	(4.88)				
EXP and GAP	YES	YES	YES				
YEAR FE	YES	YES	YES				
MVE and BTM	NO	YES	YES				
PERFORMANCE CONTROLS	NO	NO	YES				
(H3) F-stat ($\beta_1 = \beta_2$)	5.67**	5.82**	4.61**				
(H3) F-stat ($\beta_3 = \beta_4$)	3.53*	3.36*	3.57*				
Ν	898	898	893				
R^2	0.47	0.48	0.52				

Table 7 - Returns Volatility, OCI Component Volatilities, and Presentation

$$\sigma RET_{it} = \beta_0 + \beta_1 \sigma AFS_{it} + \beta_2 \sigma DERIV_{it} + \beta_3 \sigma PEN_{it} + \beta_4 \sigma FC_{it} + \beta_5 \sigma OTHER_{it} + \beta_6 PRES_{it} + \beta_7 \sigma AFS_{it} * PRES_{it} + \beta_8 \sigma DERIV_{it} * PRES_{it} + \beta_9 \sigma PEN_{it} * PRES_{it} + \beta_{10} \sigma FC_{it} * PRES_{it} + \beta_{11} \sigma OTHER_{it} * PRES_{it} + \beta_{12} \sigma NI_{it} + \sum_j \delta_j CONTROLS_{it} + \sum_{10} \gamma_{2002-2011} + \varepsilon_{it}$$
(5)

Note: The table presents the results of estimating equation (5) using the recycling sample. In Column (1), $CONTROLS_{it}$ includes derivatives exposure (EXP_{it}) and interest rate gap (GAP_{it}) . In Column (2), $CONTROLS_{it}$ includes derivatives exposure (EXP_{it}) interest rate gap (GAP_{it}) , firm size (market value of equity - MVE_{it}), and book-to-market ratio (BTM_{it}) . In Column (3), $CONTROLS_{it}$ includes derivatives exposure (EXP_{it}) interest rate gap (GAP_{it}) , firm size (market value of equity - MVE_{it}), book-to-market ratio (BTM_{it}) , comprehensive income (CI_{it-1}) , annual stock return $(ANNRET_{it})$, lagged annual stock return $(ANNRET_{it-1})$ and lagged price $(PRICE_{it-1})$. All estimations include year fixed effects, and a constant term. Control variables, year fixed effects, and the constant term are not presented for the sake of brevity. Regression coefficients are presented above t-statistics calculated based on standard errors clustered at the firm level. ***, **, and * indicate statistical significance at the 0.01, 0.05, and 0.10 levels based on two-tailed tests.

Model = PARAMETER σAFS σDERIV σPEN σFC 1 σOTHER PRES σAFS * PRES σDERIV * PRES σPEN * PRES	E (1) -2.18* (-1.75) -3.83	(2) -2.15*	(3)
PARAMETER σAFS σDERIV σPEN σFC 0FC σOTHER PRES σAFS * PRES σDERIV * PRES σPEN * PRES	(1) -2.18* (-1.75) -3.83	(2) -2.15*	(3)
σAFS σDERIV σPEN	-2.18* (-1.75) -3.83	-2.15*	1.02
σDERIVσPENσFC1σOTHERPRESσAFS * PRESσDERIV * PRESσPEN * PRES	(-1.75) -3.83		-1.95
σDERIV σPEN	-3.83	(-1.74)	(-1.60)
σPEN-1σFC1σOTHER1PRES1σAFS * PRES1σDERIV * PRES1σPEN * PRES1		-2.70	-1.81
σPEN-1σFC1σOTHER1PRES1σAFS * PRES1σDERIV * PRES1σPEN * PRES1	(-0.97)	(-0.74)	(-0.54)
σFC 1 σOTHER PRES σAFS * PRES σDERIV * PRES σPEN * PRES	11.88**	-11.87**	-12.18**
σFC 1 σOTHER PRES σAFS * PRES σDERIV * PRES σPEN * PRES	(-2.00)	(-2.04)	(-2.31)
σΟΤΗΕR PRES σAFS * PRES σDERIV * PRES σPEN * PRES	1.27**	11.92**	15.82***
$\sigma OTHER$ $PRES$ $\sigma AFS * PRES$ $\sigma DERIV * PRES$ $\sigma PEN * PRES$	(2.35)	(2.35)	(2.93)
PRES σAFS * PRES σDERIV * PRES σPEN * PRES	-2.75	-2.69	-1.25
PRES $\sigma AFS * PRES$ $\sigma DERIV * PRES$ $\sigma PEN * PRES$	(-0.44)	(-0.43)	(-0.20)
σAFS * PRES σDERIV * PRES σPEN * PRES	0.27	0.22	-0.34
σAFS * PRES σDERIV * PRES σPEN * PRES	(0.57)	(0.47)	(-0.76)
σDERIV * PRES σPEN * PRES	-0.22	-0.29	1.34
σDERIV * PRES σPEN * PRES	(-0.18)	(-0.25)	(1.18)
σPEN * PRES	0.64	-0.40	-0.61
σPEN * PRES	(0.10)	(-0.07)	(-0.09)
	-12.33	-12.41	-8.48
	(-1.34)	(-1.37)	(-1.01)
$\sigma FC * PRES$	-2.94	-0.70	-7.04
	(-0.51)	(-0.11)	(-0.83)
σOTHER * PRES	-3.75	-3.98	-4.54
	(-0.99)	(-1.05)	(-1.33)
σNI 2	2.90***	2.87***	1.82***
	(6.54)	(6.56)	(4.43)
CONTROLS	YES	YES	YES
YEAR FE	YES	YES	YES
MVE and BTM	NO	YES	YES
PERFORMANCE CONTROLS	NO	NO	YES
<i>F</i> -stat ($\beta_7 = \beta_8 = \beta_9 = \beta_{10} = \beta_{11} = 0$)	0.61	0.50	0.74
$F\text{-stat} (\beta_1 + \beta_7 = 0)$	1.67	1.82	0.15
$F\text{-stat} (\beta_2 + \beta_8 = 0)$	0.21	0.20	0.13
$F\text{-stat} (\beta_3 + \beta_9 = 0) $	7.78***	7.81***	6.86**
$F\text{-stat} \left(\beta_4 + \beta_{10} = 0\right)$	0.89	1.42	0.49
$F\text{-stat} (\beta_5 + \beta_{11} = 0)$	1.39	1.47	1.09
N			
R^2	898	898	893

Table 8 – Returns Volatility, OCI Unrealized and Recycled Subcomponent Volatilities, and Presentation

$$\sigma RET_{it} = \beta_0 + \beta_1 \sigma AFSUGL_{it} + \beta_2 \sigma AFSRECY_{it} + \beta_3 \sigma DERIVUGL_{it} + \beta_4 \sigma DERIVRECY_{it} + \beta_5 \sigma PEN_{it} + \beta_6 \sigma FC_{it} + \beta_7 \sigma OTHER_{it} + \beta_8 PRES_{it} + \beta_9 \sigma AFSUGL_{it} * PRES_{it} + \beta_{10} \sigma AFSRECY_{it} * PRES_{it} + \beta_{11} \sigma DERIVUGL_{it} * PRES_{it} + \beta_{12} \sigma DERIVRECY_{it} * PRES_{it} + \beta_{13} \sigma PEN_{it} * PRES_{it} + \beta_{14} \sigma FC_{it} * PRES_{it} + \beta_{15} \sigma OTHER_{it} * PRES_{it} + \beta_{16} \sigma NI_{it} + \sum_{j} \delta_j CONTROLS_{it} + \sum_{10} \gamma_{2002-2011} + \varepsilon_{it}$$
(6)

Note: The table presents the results of estimating equation (6) using the recycling sample. In Column (1), $CONTROLS_{it}$ includes derivatives exposure (EXP_{it}) and interest rate gap (GAP_{it}) . In Column (2), $CONTROLS_{it}$ includes derivatives exposure (EXP_{it}) interest rate gap (GAP_{it}) , firm size (market value of equity - MVE_{it}), and book-to-market ratio (BTM_{it}) . In Column (3), $CONTROLS_{it}$ includes derivatives exposure (EXP_{it}) interest rate gap (GAP_{it}) , firm size (market value of equity - MVE_{it}), book-to-market ratio (BTM_{it}) , comprehensive income (CI_{it-1}) , annual stock return $(ANNRET_{it})$, lagged annual stock return $(ANNRET_{it-1})$ and lagged price $(PRICE_{it-1})$. All estimations include year fixed effects, and a constant term. Control variables, year fixed effects, and the constant term are not presented for the sake of brevity. Regression coefficients are presented above t-statistics calculated based on standard errors clustered at the firm level. ***, **, and * indicate statistical significance at the 0.01, 0.05, and 0.10 levels based on two-tailed tests.

Dependent Variable =		σRET	
Model =		Equation (6)	
PARAMETER	(1)	(2)	(3)
σAFSUGL	-4.19***	-4.14***	-3.49**
	(-2.71)	(-2.79)	(-2.40)
σAFSRECY	5.95	5.93	5.23
	(1.55)	(1.55)	(1.37)
σDERIVUGL	-8.76*	-7.21**	-7.02**
	(-1.96)	(-1.99)	(-2.16)
σDERIVRECY	15.41	14.08	16.14
	(1.45)	(1.41)	(1.48)
σPEN	-10.45**	-10.45**	-10.89**
	(-2.03)	(-2.07)	(-2.35)
σFC	7.69	8.57	11.44*
	(1.13)	(1.23)	(1.69)
σOTHER	-4.34	-4.29	-2.83
	(-0.83)	(-0.83)	(-0.55)
PRES	0.53	0.48	-0.03
	(1.03)	(0.97)	(-0.06)
σAFSUGL * PRES	-3.87*	-3.84*	-2.21
	(-1.83)	(-1.84)	(-1.15)
σAFSRECY * PRES	7.09***	6.89***	6.47***
	(2.75)	(2.77)	(2.78)
σDERIVUGL * PRES	-14.02*	-15.54**	-8.22
	(-1.94)	(-2.33)	(-1.13)
σDERIVRECY * PRES	75.11**	77.06**	49.27*
	(2.20)	(2.31)	(1.69)
$\sigma PEN * PRES$	-9.20	-9.32	-5.69
	(-1.14)	(-1.16)	(-0.75)
$\sigma FC * PRES$	-2.93	-1.04	-4.49
	(-0.27)	(-0.09)	(-0.31)
$\sigma OTHER * PRES$	-3.12	-3.24	-3.99
	(-0.85)	(-0.89)	(-1.16)
σNI	2.61***	2.59***	1.66***
	(6.57)	(6.55)	(4.47)
CONTROLS	YES	YES	YES
YEAR FE	YES	YES	YES
MVE and BIM	NO	YES	YES
PERFORMANCE CONTROLS	NO	NO	YES
$F\text{-stat} (\beta_9 = \beta_{10} = \beta_{11} = \beta_{12} = \beta_{13} = \beta_{14} = \beta_{15} = 0)$	2.28**	2.45**	2.13**
$F\text{-stat} (\beta_1 + \beta_9 = 0)$	10.22***	10.5/***	8.71***
$F\text{-stat} \left(\beta_2 + \beta_{10} = 0\right)$	6.99***	/.18***	6.82***
$F-stat(p_3 + p_{11} = 0)$	8.91***	ð.ð/***	4.33** 5.16**
$F-stat (p_4 + p_{12} = 0)$	0.22**	0.30**	J.10** 5.04**
$P-stat(p_5 + p_{13} = 0)$	0.98 ^{***}	0.99***	5.84** 0.12
$P-stat (p_6 + p_{14} = 0)$	0.09	0.21	0.15
$r - siai (p_7 + p_{15} = 0)$	2.31	2.37	2.33
N	898	898	893
К ⁻	0.49	0.49	0.53